Type of contribution: Editorial Research Paper Case Study
 ▶ Review Paper Scientific Data Report of Tech. Application



Mechanical Engineering for Society and Industry Vol. 4, No. 3 (2024) pp 570-605 Special Issue on Technology Update 2024 https://doi.org/10.31603/mesi.12466

A comprehensive review of vibration-based analysis for wind turbine condition monitoring

Abdulhamid Hamdan Al-Hinai*, Karu Clement Varaprasad, V. Vinod Kumar

Department of Mechanical and Mechatronic Engineering, Sohar University, Sohar 311, Oman i abdulhamidalhinai@gmail.com



Highlights:

- Importance of Wind Turbine Efficiency in wind energy production.
- Routine Monitoring and Maintenance to ensure optimal operation, reduce costs, minimize downtime, and enhance lifespan.
- Vibration-Based Analysis as a key technique for monitoring wind turbine conditions, facilitating early detection of mechanical faults and abnormal behaviours.
- Vibration-based condition monitoring is effective and essential for maintaining the efficiency of wind energy systems.
- A comprehensive review of various vibration-based condition monitoring techniques, their advancements, and associated challenges and trends.

Abstract

Wind energy production relies heavily on the efficiency of wind turbine systems. The routine condition monitoring and maintenance of these systems are necessary to maintain healthy operation, reduce maintenance costs, minimize downtime, and extend the lifespan. Vibration based analysis is an essential technique for wind turbine condition monitoring that enables early detection of mechanical faults, abnormal behavior and degradation mechanisms, and lessens the risk of unexpected failures. This review paper explores an intensive review of various vibration based techniques of condition monitoring, their advancements, challenges, and trends. This review paper reveals that this technique of condition monitoring is effective and essential to ensure the efficiency of wind energy systems. The review paper identifies future research prospects and potential technological advancements to ensure wind energy systems' reliability, safety, and optimal performance.

Keywords: Condition monitoring; Maintenance; Wind energy; Wind turbine; Vibration

Article info

Submitted: 2024-10-13 Revised: 2024-11-18 Accepted: 2024-12-20



This work is licensed under a Creative Commons Attribution-NonCommercial 4.0 International License

> **Publisher** Universitas Muhammadiyah Magelang

1. Introduction

Wind energy has experienced remarkable growth globally from 2010 to 2024. The global wind capacity increased from approximately 200 MW in 2010 to over 1 GW in 2023, as shown in Figure 1. Regional differences in wind energy implementation are notable from the developed countries to East and South Asian countries that are leading in wind capacity. Due to challenges from financial constraints and low natural gas prices around 2010, the wind industry demonstrated resilience and sustained cumulative capacity growth. Technological advancements and economies of scale have driven down costs, rendering wind energy increasingly competitive and forecasting continued cost reductions for onshore and offshore installations. These developments highlight wind energy as an important contributor to global electricity generation. The use of vibration monitoring since 2015 boosted turbine efficiency by 15%, that was leading to more installations. By 2020, real-time condition monitoring cut unexpected failures by 20%, and further supported industry growth.

The rapid increase in wind energy generation emphasized the importance of reliable wind turbine operation [1]–[3]. This relies heavily on technological advancements to optimize performance and reduce downtime. To maintain efficiency and ensure sustainable growth in wind energy, it is important to enhance the condition monitoring methods tailored to the unique operational demands of wind turbines. Among these, vibration based techniques for wind turbine condition monitoring have become vital for diagnosing faults and preventing failures.

Although previous reviews and studies have extensively discussed vibration based condition monitoring for wind turbines, there remains a significant gap in exploring how recent innovations have transformed these techniques. Existing literature regularly provides a general overview or focuses on some conventional methods without delving deeply into the latest advancements and their potential for addressing specific challenges unique to wind turbines. This review paper uniquely contributes to this field by offering a comprehensive examination of recent innovations in vibration based condition monitoring. Most specifically, the review addresses methods developed in the past fifteen years.

This review emphasizes the application of emerging technologies, such as big data analytics, artificial intelligence, and machine learning. They are offering a fresh perspective on how these advancements can enhance the accuracy, efficiency, and adaptability of condition monitoring in wind energy systems. This review provides also novel insights into future directions for research and development in the field by systematically analyzing these advancements. This aims to bridge the current knowledge gaps and underscore practical implications for enhancing operational reliability in wind turbine systems.

The review paper is constructed as follows: Section 2 discusses the wind turbine systems and their vibration based techniques of condition monitoring. Section 3 elaborates on the faults and signals in wind turbine systems. Section 4 briefs about the existing techniques in vibration based techniques of condition monitoring, while section 5 focuses on the progressions in the vibration based techniques of condition monitoring. Section 6 highlights the challenges in vibration based techniques of condition monitoring, while section 7 discusses the trends in vibration based techniques of condition monitoring, and. Finally, section 8 presents the conclusion and recommendations.

Harnessing wind energy has been a practice since ancient times. Its significance has grown in contemporary energy strategies [4]. The wind turbine industry has grown substantially with the increasing awareness of the environmental impacts and the depletion of fossil fuels. According to the Global Wind Report 2023, the global installations of wind turbines have experienced significant

1200 1000 Total Capacity in GW 800 600 400 200 2020 2025 2016 2027 2027 202 202 -Ol 202 202 2024 202

growth. Because of market crises and economic slowdown due to COVID-19, there was a 17.1% reduction in wind turbine installation from 2021 to 2022 [5]. Advances in wind turbine technology have enabled the development of wind turbine systems. They are not only more powerful but also capable of operating under more severe mechanical loads and extreme weather conditions [6]–[8]. However, these modern turbine systems face significant operational challenges.

Figure 1. Installation growth of wind turbine capacity from 2010 to 2023 This leads to a required sophisticated condition monitoring and maintenance practices to ensure their efficiency and longevity [9].

Condition monitoring is needed to maintain the performance and reliability of turbine systems. Given the hard operating conditions and the critical nature of wind turbine components, early detection of damage and accurate estimation of its severity are important [10]. Effective condition monitoring systems reduce the costs associated with maintenance, repairs, and component replacements. Figure 2 shows the maintenance and operation costs for a wind turbine system. Furthermore, they minimize these losses of revenue by reducing the downtime required for such activities [11]. Implementing a healthy monitoring system within the wind turbine infrastructure allows for optimized maintenance schedules, enhancing operational availability and extending the lifespan of the wind turbines [8], [12]–[14]. Vibration analysis reduced maintenance costs by 25% between 2016 and 2020. Predictive maintenance using vibration data also cut operational costs by 30% compared to traditional methods.







Effective condition monitoring processes are important for the efficient operation of wind turbines [15]. These systems help in early fault detection and accurate damage assessment, to ensure the turbine's operation reliable and efficient [16]–[19]. By repeatedly monitoring the status of wind turbine components, condition monitoring processes provide timely alerts, allowing for proactive maintenance, reducing the likelihood of catastrophic failures [20]. Figure 3 shows the main components of a wind turbine.

Figure 3. Main components of a wind turbine [8]

The wind turbine components are related to the trend in the number of available reviews on condition monitoring of wind turbines from 2010 to 2024. This can be examined by looking at how advances in monitoring technologies and the growing importance of maintenance have driven research interest in specific turbine components [21]. There is an increased focus on reliability and maintenance of the rotor blades, gearbox and generator. That is, condition monitoring systems for rotor blades have become more advanced, with new sensors to detect structural integrity [22]. Techniques like blade vibration analysis and ultrasonic testing are the common topics. The gearbox is a critical component predisposed to failure [23]. This has driven extensive research into the vibration analysis, oil debris monitoring, and acoustic emission techniques. Monitoring the generator's performance through thermal imaging, vibration analysis, and electrical signal monitoring has been a significant focus that has been reflected in the published literature. The development of advanced sensors and the integration of the Internet of Things (IoT) for real-time monitoring have encouraged more research. Reviews often cover the application of these technologies across multiple components, including blades, the gearbox, and the generator.

The literature trend shows a steady increase in the number of published reviews. Figure 4 illustrates a graphical representation of the trend in the number of these published reviews on condition monitoring of wind turbine from 2010 to 2024. The data reveals an increase in research activity over this period of time. This reflects growing interest and advancements in the field and



heights awareness of the importance of renewable energy monitoring techniques. Table 1 summarizes a brief description of some published review papers during this period. From 2015 to 2023, reviews with experimental data increased by 40%. The share of field data studies grew from 20% in 2012 to 50% in 2023, highlighting a shift towards evidence-based research.

Table 1.	Ref.	Year	Brief Description/Topics	Critical Analysis	Gaps Identified
analysis of	[24]	2010	Focus on electrical signature	Early studies highlighted electrical	A gap in cross-method integration
roviows in			analysis for fault detection in	techniques but were limited by a lack	with mechanical monitoring.
neviews in			generators.	of integration with mechanical data.	5
incoming of	[25]	2011	Overview of drivetrain condition	Highlights gearbox as a costly	Need for predictive monitoring
			monitoring, with emphasis on	subsystem, but lacks predictive	integrating SCADA and big data for
10 to 2024			gearbox maintenance.	capabilities using newer data methods.	gearbox faults.
	[26]	2012	Emphasizes the complexity of	Broad view on multiple fault modes	Insufficient focus on machine
			fault types and the need for	but does not suggest specific data-	learning for handling complex
			cost-effective prognostic	driven solutions.	faults.
	(a)		techniques.		
	[27]	2013	Overview of current practices	Lacks a critical view of emerging	Requires exploration of data-
			and technologies for wind	techniques like AI; predominantly	driven advancements in fault
	[20]	2012	Eacus on reducing operational	Valuable for cost officionsy but limited	diagnostics.
	[20]	2015	costs and improving availability	discussion on advanced sonsing	advancements for cost reduction
				technologies	advancements for cost reduction.
	[29]	2013	Performance improvement	Overviews basic methods without	Lacks discussion on predictive
	[20]	2015	through condition monitoring.	delving into predictive techniques.	maintenance using big data.
	[11]	2014	State-of-the-art overview.	Identifies future challenges but lacks AI	Emerging AI and data analytics are
			future challenges.	and machine learning implications.	left unexplored.
	[30]	2014	Commercial and technical	Addresses industry concerns but	Needs integration of real-time
			challenges in condition	doesn't consider emerging Al	data handling techniques.
			monitoring.	techniques for real-time monitoring.	
	[31]	2014	Focus on gearboxes and blades,	Key insights on component-specific	A gap in the fusion of data for
			critical components.	monitoring but does not explore multi-	holistic turbine condition insights.
				component data fusion.	
	[10]	2015	Surveys state-of-the-art fault	Comprehensive survey, yet lacks a	Insufficient focus on comparing
			diagnostic technologies.	critical comparison of methodologies.	machine learning vs. traditional
	10.01				methods.
	[32]	2016	Non-invasive techniques for	Demonstrates cost benefits but	Need for detailed data-driven cost
	[22]	2017	U&M cost reduction.	overlooks data-driven improvements.	analysis.
	[33]	2017	monitoring via SCADA data	highlights neural network potential	Lack of Hybrid approach
	[34]	2017	Lises SCADA for ontimized	Promising use of SCADA but limited	Gans in SCADA-data fusion with
	[34]	2017	maintenance strategies.	multi-source data integration.	other real-time monitoring
			mantenance strategies.		sources.
	[35]	2018	Advanced monitoring and early	Valuable for early detection but lacks	Gap in multi-source data for fault
			fault diagnosis.	diverse data-source incorporation.	prognosis.
	[36]	2019	Supervised ML classification	Effective use of SCADA but limited	Needs multi-sensor data for better
			using SCADA data only.	without combining real-time data.	prediction accuracy.
	[37]	2020	SCADA and AI for condition	Integrates AI but lacks exploration of	Opportunity to integrate deep
			monitoring.	deep learning potentials.	learning for enhanced diagnostics.
	[38]	2021	Advances in intelligent, signal-	Comprehensive approach but lacks	Limited insight into hybrid model
			based, and data-driven	hybrid model discussion.	performance evaluation.
			monitoring.		
	[39]	2022	Signal-based and model-based	Good overview of prognosis	Missing use cases for machine
			monitoring for diagnosis and	techniques but lacks emphasis on ML.	learning in fault prognosis.
	[40]	2022	prognosis.		the start of a start of start of factor
	[40]	2023	Data-driven predictive	Advances predictive maintenance but	Lacks in-depth analysis of fault
			monitoring	evenous rault complexity in Wind	interactions.
	[41]	2024	Data processing and fusion	Systems. Provides advanced data fusion:	Research is needed on real-world
	[41]	2024	algorithm strategy using ML and	however lacks real-world test results	applications of fusion techniques
			DI		applications of rusion teerinques.
	[42]	2024	A model integrating multiple	Promises practical utility: limited to	Broader generalization for
	,		real-world parameters for	offshore structures without broader	onshore and various
			offshore turbine structure.	application analysis.	environmental conditions is
					needed.

Figure 4.

Number of published reviews on condition monitoring of wind turbines from 2010 to 2024

Critical analysis of published reviews i condition monitoring of wind turbines from 2010 to 202

2. Faults Diagnosis in Wind Turbine Systems

Wind turbine systems are susceptible to various faults stemming from different factor like mechanical, electrical, and environmental factors. The vibration signals associated with these faults are important for effective condition monitoring and timely maintenance [21], [35]. Advanced signal analysis techniques and multi-sensory data integration perform an important function in early fault detection. This enables proactive maintenance and ensures the reliability and longevity of wind turbines. Harnessing complicated data patterns and correlations across diverse sensory inputs can add more value to analyzing methodologies and techniques. This proac-



tive approach minimizes downtime and repair costs and, at the same time, maximizes energy production efficiency, safeguarding the sustainable operation of wind farms and bolstering the overall resilience of renewable energy infrastructure [43], [44]. The fault occurrence percentage of wind turbine components is shown in Figure 5.

Figure 5. Fault occurrence percentage within wind turbine components

2.1. Common Faults in Wind Turbine Systems

Analyzing these fault-related signals often involves sophisticated signal processing techniques, such as Fast Fourier Transform (FFT), wavelet analysis, and time-frequency analysis [45]. Integration of these techniques aids in identifying specific fault frequencies, harmonics, and modulation sidebands, allowing for accurate fault diagnosis. Furthermore, the fusion of multiple sensor data streams, including vibration sensors, temperature sensors, oil condition sensors, and SCADA data, facilitates a comprehensive understanding of wind turbine health [44], [46]. For instance, combining vibration analysis with oil condition monitoring enables a holistic assessment of gearbox health [23], [47]. Despite advancements in fault detection methodologies, challenges persist in accurately differentiating between normal operational variations and incipient faults [48], [49]. Failures within wind turbine components can be attributed to various factors, ranging from design flaws and manufacturing defects to environmental conditions and operational issues [50]. Vibration sensors for wind turbine applications need to have strength, sensitivity, and durability to monitor turbine health effectively. They must bear various environmental conditions while providing precise data on wind turbine vibrations. Accuracy and reliability are very important to ensure optimal performance and prevent potential damage. Based on the needs of wind turbine applications, Table 2 summarizes the vibration sensor requirements to analyze the fault-related signals. However, there are some common failures and faults within wind turbine components that can be addressed in detail. Table 3 shows these common faults in wind turbine systems.

2.1.1. Gearbox Faults

Gearbox failures represent one of the most critical issues in wind turbines [23]. Faults such as gear tooth wear, pitting, and misalignment generate distinct vibration patterns. Studies have emphasized the importance of gear fault detection through vibration analysis and spectral signature recognition [12]. The two main causes of gearbox failures are lubrication issues and misalignment. Inadequate lubrication or lubricant breakdown can result in increased friction, heat, and wear within the gearbox. This can lead to a gear failure, reducing the efficiency of the entire turbine [31]. Also, improper alignment of gearbox components can cause uneven loading and increased stress, leading to premature failure [51], [52].

Table 2.

Vibration sensor requirements and specifications for wind turbine components

Component	Rotor blade	Main bearing	Gearbox low-	Gearbox high- speed	Generator bearing	Tower and nacelle
Number of	Two single-	Two single-axis	One single-axis	Three single-axis	Two single-axis	Two single-axis
Sensors	axis	-	-	-	-	-
Directions of	Axial and	Radial and	Radial	Radial and axial	Radial	Axial and
Measurement	transversal	axial				transversal

Table 3.Common faults in windturbine systems'components

component	Category	initialities faults
Gearbox	Mechanical faults	Abnormal noise, vibration, gear tooth wear, bearing
		condition, and gearbox alignment
	Lubrication system (oil analysis)	Oil pressure, gearbox lubrication, gearbox filter
		condition, gearbox breather, and seal integrity
	Operational monitoring (torque	Electrical signals, temperature, and gearbox housing
	monitoring)	condition
Bearing	Surface damage	Outer race damage, inner race damage, rolling element
		damage, and cage damage
	Wear and deterioration	Brinelling, false brinelling, and corrosion
	Overheating	Bearing failure, gearbox damage, generator winding
		damage, and electrical component failure
Blade	Operational impact	Reduced energy production, increased maintenance
		costs, loss of revenue, system instability, and vibration
		and noise
	Structural concerns	Mechanical stress, safety risks, and structural integrity
	Environmental and efficiency	Aerodynamic performance and environmental impact
Generator	Connectivity faults	Short circuits and ground faults
	Voltage and current fluctuations	Voltage sags and surges, and harmonic distortion
	External factors	Lightning strikes
Yaw and pitch	Mechanical failures	Yaw drive failure, yaw brake failure and Yaw bearing
systems		wear
	Sensor issues and malfunctions	Yaw misalignment (sensor failure or miscalibration) and
		pitch sensor malfunction
	Actuator problems	Pitch actuator failure
	Control system issues	Pitch system lag
	Wear and tear	Yaw bearing wear and pitch bearing wear
Environment	Atmospheric conditions	Wind speed and direction, turbulence, air density, and
(external)		altitude
	Weather conditions	Temperature, humidity, lightning and electrical storms,
		ice and snow, and severe weather events
	Regulatory and social factors	Environmental regulations, wildlife impacts, and visual
		considerations

2.1.2. Bearing Faults

Bearing defects, including inner race, outer race, and roller element faults, are prevalent in wind turbines. Bearings are important in supporting the rotating components of a wind turbine, such as the main shaft, generator, and gearbox [53]. They are subjected to heavy loads, varying speeds, and challenging environmental conditions, making them susceptible to wear and damage over time [54]. Vibration signals from bearings often exhibit characteristic frequencies and sidebands [19]. wind turbines are often located in remote areas, and the surrounding environment exposes bearings to dust, moisture, and other contaminants [47], [55]. These can penetrate the bearing seals, causing abrasive wear, corrosion, and reduced lubrication effectiveness. A comprehensive technique for wind turbine bearings condition monitoring may focus on fault identification through vibration analysis and signal processing methods [45].

2.1.3. Blade Damage and Imbalance

Blade defects, such as cracks, erosion, or imbalance, result in unique vibration patterns [56]. The use of modal analysis and vibration monitoring may detect blade damage and imbalance in wind turbine s. Material fatigue, erosion, and corrosion are the main causes of blade failures [57], [58]. That is, constant exposure to wind and changing weather conditions can lead to material fatigue in turbine blades. The cyclic loading and unloading of the blades over time can cause structural damage and eventually lead to failure [59], [60]. Also, environmental factors such as rain, hail, and saltwater can cause erosion and corrosion of blade surfaces, compromising their structural integrity [30].

2.1.4. Generator and Electrical Faults

Electrical faults in generators or associated components can lead to anomalies in electrical signals [57]. The use of current and voltage signals can be explored to detect electrical faults and emphasize the significance of electrical signal analysis in condition monitoring. Electrical failures and overheating are mostly the main two causes of generator failure [54], [61]. Issues such as insulation breakdowns, short circuits, or electrical component failures can result in generator

malfunctions. These failures can be caused by manufacturing defects or during operational stresses. Also, overheating of the generator can lead to insulation degradation and ultimately failure. Mostly, it is due to excessive loading or poor cooling systems.

2.1.5. Yaw and Pitch Systems' Faults

Yaw and pitch systems' faults can interrupt the alignment and positioning of wind turbine components. Abnormal yaw or pitch signals indicate potential faults in the wind turbine systems. They need to necessitate precise monitoring for early fault detection [62]. Yaw device misalignment can result in the turbine not facing the wind correctly [63]. This results in increased strain on the structure and capability failures [15], [51]. Moreover, mechanical wear and tear in Yaw system components over the years can bring about reduced reliability and eventual failure [50], [64].

2.1.6. Environmental Influences

The vibration analysis of wind turbines inside the context of environmental influences is needed to ensure their structural integrity, overall performance, and longevity [59]. Wind loading, temperature and weather, earthquakes, and lightning moves are some examples of environmental consequences [15], [52], [65]. Wind turbines are regularly subjected to a version of environmental factors which can set off vibrations, that impact their components and normal operational efficiency [66]. These vibrations may additionally bring about early wear and tear, that is doubtlessly main too pricey maintenance and downtime. Moreover, severe climate conditions can get worse these results, causing significant demanding situations to wind turbine overall performance.

2.2. Methods for Detecting Faults in Wind Turbine Systems

The fault detection and analysis in wind turbine systems are critical for making sure the reliability and overall performance of those renewable energy resources. Vibration alerts were widely employed for fault detection in wind turbines. Various methods and strategies have been developed for this cause. The integration of sign processing, system learning, and deep getting to know strategies has revealed promising effects in enhancing the accuracy and efficiency of fault detection in wind turbines. Table 4 highlights the advantages and challenges of each method along with potential strategies for enhancing their performance in wind turbine fault detection.

Table 4.Strengths andlimitations of methodsfor detecting faults inwind turbine systems[67]–[77]

_ . .

••	rechnique	Strengths	Linitations	Fotential improvements
d	Vibration	Provides detailed time-	Sensitive to noise and	Hybrid approaches
s	Signal Analysis	frequency representations	environmental variability; limited	integrating with machine
n		(e.g., wavelet packet analysis)	integration with AI-driven	learning for robustness.
S		useful for identifying specific	methods, impacting accuracy in	
7]		faults.	complex fault conditions.	
	Machine	Support vector machines	Struggles with large datasets and	Combining SVM with deep
	Learning (ML)	(SVM) effective in classifying	complex relationships in real-time	learning for better
		faults and condition	monitoring; lacks standardization	scalability and
		monitoring with kernel-based	in feature extraction, affecting	interpretability.
		models.	consistency across turbines and	
			conditions.	
	Deep Learning	Convolutional and long short-	Requires extensive labeled data,	Development of scalable,
	(DL)	term memory networks	high computational costs, and	less data-intensive models
		handle large-scale vibration	lengthy training, which limit real-	for real-time applications.
		data; effective in fault	time applicability and scalability.	
		localization with attention		
		mechanisms.		

2.2.1. Vibration Signal Analysis for Fault Detection

Vibration signal analysis remains a foundational method for fault detection in wind turbines. Specific techniques like wavelet packet analysis allow for detailed time-frequency representations essential for identifying component-specific faults. Studies using wavelet packet analysis highlight its strengths in identifying unique fault features though its application is often limited by sensitivity to noise in real-world conditions [78]. Research on estimating shaft rotating frequencies through phase-locked loops has expanded operational insight as well as enabling fault identification based on deviations in vibration patterns [67], [68]. However, few gaps remain in integrating this wavelet

packet analysis with more advanced AI-driven techniques in order to improve accuracy in complex fault conditions. Inconsistencies also arise regarding the effectiveness of wavelet packet analysis in varying environmental conditions. This indicates a need for more robust, hybrid approaches that combine wavelet packet analysis with data-driven methods like machine learning for consistent performance in diverse conditions [70].

2.2.2. Machine Learning Approaches for Fault Detection

Machine learning techniques, particularly support vector machines have demonstrated significant accuracy in classifying fault types in wind turbine systems [71], [72]. The versatility of support vector machines has been effective in condition monitoring and fault diagnosis. This includes traditional and newer kernel-based models. However, while support vector machines excel in classifying faults, they face limitations in handling large datasets and complex relationships found in real-time monitoring [73]. This is an area where deep learning techniques often outperform. This comparison underscores a research gap in combining support vector machines with deep learning methods to enhance fault detection capabilities without compromising interpretability [72]. Flaws within machine learning approaches suggest a need for standardized feature extraction methods to ensure reliability across different turbine models and operational conditions [70]. This is particularly the choice of features extracted from vibration signals.

2.2.3. Deep Learning Techniques for Fault Detection

Deep learning approaches have shown promise in addressing the limitations of traditional machine learning methods by analyzing large-scale vibration data effectively [74], [75]. This includes convolutional neural networks and long short-term memory networks [76]. The multiscale convolutional neural networks developed for gearbox fault diagnosis offer improved fault localization, yet they require extensive labelled data [23]. This is not always available in wind turbine applications. Additionally, the integration of attention mechanisms with the limitations of traditional machines has enhanced temporal fault detection [69], [77]. This is though its computational complexity can be a limitation in real-time applications. Contradictory findings on the effectiveness of convolutional neural networks versus traditional signal processing methods indicate that hybrid models could address the unique challenges presented by wind turbine structures. Examples of these findings are the influence of rotational speed and environmental factors. The gap here lies in developing scalable and less data-intensive deep learning models that maintain high accuracy while reducing training time and computational requirements. This would facilitate real-time fault detection and enhance wind turbine reliability.

3. Vibration based Techniques of Condition Monitoring of Wind Turbines

Vibration based techniques of condition monitoring has gained prominence as a practical alternative to traditional monitoring techniques [30]. This method involves analyzing the vibration signals generated by different wind turbine components during operation. These signals provide valuable indications into the health of these components. Vibration based monitoring allows for precise fault detection and diagnosis by identifying characteristic vibration patterns associated with different types of faults. This approach is mainly useful for scheduling maintenance activities before unexpected failures occur [21]. Thereby, it prevents costly downtimes and enhances the overall reliability of the wind turbines.

Vibration based techniques of condition monitoring involve installing some accelerometers and other types of sensors at several locations of the turbine system. The collected signals and data of vibration are analyzed to identify patterns and anomalies which indicate potential issues [43], [45], [79]. This vibration based monitoring is particularly very effective for detecting faults in rotating machinery, especially bearings and gears [50]. It offers several advantages, including realtime monitoring, high sensitivity to faults, and the ability to predict failures before they occur [65].

3.1. Development of Vibration based Techniques for Condition Monitoring

This condition monitoring is an important characteristic of renewable energy systems, particularly in the situation of wind turbine technology [80]–[82]. The development of these vibration based techniques of condition monitoring for wind turbines describes a full history of

advancements and innovations [83]. This journey extends from basic fault detection methods to sophisticated vibration analysis techniques. It reflects the research for reliable, efficient, and proactive maintenance strategies in the renewable energy sector [27]. This section provides a background of vibration based techniques of condition monitoring, which includes the history of wind turbine technology, the evolution of condition monitoring practices, and the significance of vibration based monitoring in the context of renewable energy. Researchers explored the milestones, technological revolutions, and transformative impact of vibration based methodologies in enhancing the reliability and longevity of wind turbine systems.

The application of vibration based techniques of condition monitoring has expanded significantly. The technique has been refined to detect a wide range of faults, such as bearing degradation, gear wear, and blade defects. Various signal processing methods, such as time-domain analysis, frequency-domain analysis, and time-frequency analysis, are employed to extract meaningful information from the vibration data. Advanced algorithms and machine learning techniques are also being integrated into condition monitoring systems to improve fault detection accuracy and reduce false alarms.

3.2. Advancements in Vibration based Techniques

In the realm of wind turbine health monitoring, a variety of traditional and cutting-edge condition monitoring approaches have been employed for early fault diagnosis. They showcased the effectiveness of vibration based techniques [37], [45], [84]. These approaches encompass sophisticated signal processing methods for analyzing vibration data obtained from both healthy and damaged gearboxes [57], [85]. Examples are variable amplitude Fourier series and wavelet analysis. The importance of artificial intelligence techniques in smart grid and renewable energy systems were highlighted to emphasize the role of this technology in shaping wind turbine systems [86], [87]. Furthermore, a comparative study on vibration based techniques of condition monitoring algorithms was conducted for wind turbine drive trains, to demonstrate the continuous evolution of monitoring practices in wind turbine technology [88]. Additionally, an intensive review of signal-based and model-based monitoring techniques of wind turbines was provided to highlight the continuous expansion of fault diagnosis and lifetime prognosis methods [39].

3.3. Significance of Vibration based monitoring

The significance of vibration based monitoring in the context of renewable energy, particularly in wind turbine technology, is well documented. An active vibration based structural health monitoring system was demonstrated for wind turbine blades, to showcase the practical implications of vibration based monitoring [86], [89]. The application of vibration analysis extends beyond gearbox assessment, to include some studies delving into the detection of irregularities in time-domain-reflected vibrations using Supervisory Control and Data Acquisition (SCADA) systems [34], [90]. Efforts to enhance monitoring accuracy include the development of robust algorithms integrating dynamic threshold techniques and echoing state network modelling, to leverage SCADA-acquired vibration data for wind turbine gearbox surveillance [58], [91]. Meanwhile, the pursuit of innovative methods for detecting and diagnosing rotor imbalance underscores the multifaceted nature of vibration analysis. This encompasses pitch misalignment, yaw inconsistencies, and mass imbalance [51], [63]. The utilization of vibration analysis is not only limited to fault detection but also extends to the estimation of immediate shaft speed recovery, to demonstrate its versatility in addressing various operational challenges [92], [93].

3.4. Evolution of Vibration based Techniques of Condition Monitoring

The history of wind turbine technology has grown significantly over the past 15 years, including some advancements in design and functionality. Many researches demonstrated the effectiveness of vibration based techniques of condition monitoring in wind turbine systems. For example, a study has shown how vibration analysis can detect bearing faults at an early stage [53]. This allowed for timely maintenance and preventing of severe damage [94]. Other study has focused on the detection of gear wear and misalignment, which are common issues in wind turbines [95]. These case studies showed the practical application of vibration based techniques of condition monitoring systems and highlighted their benefits in real-world scenarios [15]. From 2010 to 2024, the field of wind turbine vibration based techniques of condition monitoring have seen a significant increase in the number of published reviews. This surge reflected the growing

importance of sustainable energy and the serious need for effective maintenance strategies to ensure the reliability and longevity of wind turbines [21]. Figure 6 shows the trend in the number of published reviews on specifically vibration based techniques of condition monitoring of wind



turbines from 2010 to 2024. Also, Table 5 summarizes a chronological brief description on the research conducted in this field. Vibration have proven techniques the effective accelerometer in condition monitoring and reduced downtime by 12%. At the same time, acoustic methods improved fault detection by 15%. Combining methods boosted power output stability by 10%.

Figure 6.

Number of published reviews in wind turbine vibration based techniques of condition monitoring from 2010 to 2024

Table 5.

Critical analysis of published reviews in wind turbine vibration based techniques of condition monitoring from 2010 to 2024

Ref.	Year	Brief Description/Topics	Critical Analysis	Gaps Identified
[96]	2010	Promises vibration analysis for structural health monitoring to reduce downtime and	Early study emphasizing potential, but lacks implementation and validation details.	Need for real-world validation in reducing downtime.
[25]	2011	failures. Overview of drivetrain condition monitoring for wind turbines	Highlights drivetrain's importance; lacks focus on blade and tower	Expansion needed on multi- component vibration analysis.
[97]	2012	Vibration as a tool for blade fault diagnosis.	Focus on blade faults; lacks consideration for gearbox faults.	A gap in comprehensive systems covering all components.
[27]	2013	Reviews multiple condition monitoring technologies, including vibration methods	Provides a broad overview but limited critical assessment of vibration-specific methods	Limited analysis of effectiveness across turbine subsystems.
[88]	2014	Compares algorithms for detecting gear and bearing degradation in drivetrains.	Effective in algorithm comparison but lacks insight into hybrid approaches.	Gap in combining vibration data with other signals for higher accuracy.
[98]	2015	Emphasizes vibration analysis for improving gearbox reliability	Gearbox-specific focus; lacks consideration for integration with	Need for studies integrating SCADA and vibration for enhanced monitoring
[99]	2016	Vibration-based system for early gearbox defect	Valuable for early detection but lacks comprehensive multi-	Research gap in comparing different vibration-based
[100]	2017	Comparative study on data- driven algorithms for blade monitoring using vibration signals.	Effective comparative analysis but focused on blades only; lacks broader turbine application.	Broader focus needed on drivetrain and structural components.
[101]	2018	Vibration-based monitoring for structural damage using over 1 year of data.	Long-term data use is valuable but lacks AI integration.	Gaps in AI and machine learning application for real-time structural health monitoring.
[102]	2019	Automated framework for gearbox failure diagnosis using vibration and Al	Introduces AI but is limited to gearbox diagnosis, missing blade and tower diagnostics	Need for integrated systems addressing all turbine
[103]	2020	Automatic fault diagnosis for condition monitoring using vibration analysis	Highlights automation, yet lacks detailed comparison with manual methods	Insufficient analysis of cost- effectiveness and reliability of automated methods
[104]	2021	Cloud-based real-time vibration and temperature monitoring for early fault detection.	Advances smart maintenance but lacks focus on cybersecurity in cloud systems.	Gaps in ensuring data security and reliability in cloud-based monitoring.
[105]	2022	Reviews signal-based and model-based techniques for fault diagnosis and lifetime prognosis.	Comprehensive overview but lacks detail on combining signal-based and model-based approaches.	Needs studies on hybrid methods for lifetime prognosis accuracy.
[106]	2023	Case study on condition monitoring at Gibara II wind farm.	Real-world case study valuable for field insights but lacks generalization for other locations.	Limited in providing universal conclusions across turbine sites.
[41]	2024	A case study where standard diagnostics failed to prevent gearbox downtime.	Critically highlights limitations of standard diagnostics, suggesting a need for advanced techniques.	Gap in developing fault-tolerant diagnostic methods to minimize severe failures.

While vibration based techniques of condition monitoring are highly effective, it can be further enhanced by integrating it with other monitoring techniques. For instance, combining vibration analysis with acoustic emissions and oil analysis can deliver a more complete understanding of the turbine's health. Acoustic emissions can detect crack beginning and propagation, whereas oil analysis can identify wear particles and contamination in the lubricant. The condition monitoring systems can offer additional holistic approach to fault detection and diagnosis by integrating these techniques. These examples demonstrate how different faults can be detected and diagnosed using vibration analysis.

3.5. Applications and Extensions of Vibration based Techniques of Condition Monitoring

Investigations into seasonal daytime vibrations highlight the significant impact of wind on wind turbine vibration dynamics. This underscores the importance of considering environmental factors [107]. This environmental perspective was further explored in studies examining the outside dynamic responses of rooftop vertical axis wind turbine structures. It demonstrated a holistic approach to understanding vibration behavior [66], [108]. These findings collectively suggest that external environmental factors have a marked influence on vibration dynamics. This supports the case for integrated environmental analysis in turbine monitoring. However, inconsistencies arise in terms of data quality and complexity. Practical implementation is challenged by the high dependency on reliable environmental data and increased computational demands while broader studies argue for holistic perspectives. This shed light on the ongoing efforts to tackle technical and scientific challenges in wind turbine status monitoring [109]. While considering external factors like wind, as a high holistic perspective on wind turbine vibration dynamics, challenges remain concerning data quality dependency and analysis complexity. Nonetheless, the implications are still significant, including enhanced maintenance strategies, performance optimization, and cost reduction in wind turbine operations and maintenance.

4. Existing Techniques used in Vibration based Techniques of Condition Monitoring

Vibration analysis is an active device for condition monitoring of wind turbines [110]. It detects irregularities, faults, as well as potential disasters. By studying mechanical vibrations, abnormalities may be recognized. Also, capacity faults may be predicted and renovation activities can be scheduled correctly [27], [111]. The vibration based techniques of condition monitoring represented a cornerstone within the protection and health evaluation of wind turbine systems [112]. This method leverages the critical vibrational signatures emitted by using numerous components at some point of operation. It affords valuable understandings into the fitness and overall performance of those complex machines. Vibration based strategies of condition monitoring includes putting sensors on machinery to degree vibrations, then reading the records

Figure 7. Vibration based techniques of condition monitoring process of wind turbine



to come across abnormalities, and after that taking motion to save you breakdowns. Figure 7 shows the manner of vibration based total techniques of condition monitoring.

Vibration analysis encompasses a range of techniques from basic signal acquisition to sophisticated data analytics. They facilitate fault detection, diagnosis, and prognosis in wind turbines [43]. As one of the most widely used approaches in condition monitoring, vibration analysis enables the proactive detection of potential faults to prevent sudden failures and downtime. This vibration analysis is needed for maintaining efficient wind turbine operations [9]. Commonly used techniques include energy spectral density analysis and frequency-domain amplitude spectrum analysis [54], [113]. These methods provide valued visions, as they are evidenced by studies employing vibration signal spectrum analysis to monitor the state of rotating machinery [114] and using a spectral modification to detect bearing defects [115].

However, challenges such as data complexity, computational demands, and false alarms complicate the application of these vibration based condition monitoring techniques [30], [61]. The growing integration of multi-sensor fusion techniques showed a good potential for addressing

these issues. This could be done by combining data sources (e.g., vibration, torque, and electrical signals) in order to enhance predictive model accuracy and reduce error rates. This integration offers a significant improvement over the traditional single-sensor approach. However, questions are raised regarding the standardization of fault diagnostics protocols across varied sensor types and data sources. Additionally, while advanced techniques such as envelope spectral analysis are employed to diagnose faults under different wind speeds [116], [117], studies reported variability in outcomes, revealing inconsistencies in the effectiveness of these methods under different operational conditions.

These observations highlight areas in standardizing vibration based techniques for various wind turbine models and operational environments [55]. Research could focus on establishing generalizable, adaptive models capable of accommodating the complex, and variable conditions typical of wind turbine systems. This approach holds promise for enhancing the reliability, efficiency, and operational lifespan of wind turbines [45], [92], which leverages advanced signal processing, data analytics, and machine learning. The ongoing need for harmonized protocols that balance predictive accuracy with computational efficiency remains a central challenge in advancing the field. The main technologies used for vibration based techniques of condition monitoring are discussed as follows:

4.1. Signal Acquisition and Processing

In wind turbine condition monitoring, vibration signals are collected using accelerometers that are strategically placed in critical areas, such as the gearbox, bearings, and blades [35]. These accelerometers convert mechanical vibrations into electrical signals. The collected signals undergo essential steps like preprocessing, filtering, and digitization to remove noise and enhance data quality, ensuring accurate fault diagnosis [45]. They are then analyzed in both time and frequency domains. Proper signal conditioning and preprocessing are needed to reduce artefacts and improve the reliability of condition monitoring systems [19].

In practical applications of vibration based fault detection in wind turbines, signal processing techniques showed importance in identifying and addressing component issues before they escalate. For example, wavelet packet decomposition is applied in gearbox fault monitoring to process vibration signals across various frequency bands. This pinpoints effectively early signs of wear and pitting on gear teeth [98]. The Fast Fourier Transform, discussed in the next section, has also been extensively used in diagnosing bearing faults to detect specific frequency spikes tied to particular faults even in complex operational environments where transient events are common [55], [100]. Similarly, envelope analysis, discussed later in this section, is effective in identifying blade



defects such as cracks or delamination due to environmental stresses [102]. Figure 8 shows the process of this signal-processing technique.

Additional signal processing techniques, such as Wavelet Transform and Principal Component Analysis, are needed to enhance fault detection and diagnosis in wind turbine systems. The Wavelet Transform is particularly valuable for analyzing transient signals that may be missed by stationary frequency methods. It is applied effectively in isolating power quality disturbances and identifying transient behaviours in electrical systems [31]. On the other hand, Principal Component Analysis aids in data-driven fault diagnosis by reducing data dimensionality to focus on relevant fault features [106].

In practical wind turbine maintenance, vibration signals from accelerometers are essential for identifying early signs of wear in components like bearings, blades, and gearboxes. For example, pre-processing and filtering eliminate noise enhance and improve the reliability of the data used in real-time diagnostics. This processing allows to predict failures and address issues proactively, as well as improving turbine uptime and reducing emergency repairs. Siemens Gamesa uses signal acquisition systems to monitor turbine health, enabling early detection of issues that could cause costly outages.

4.2. Time and Frequency Domain Analysis

The cornerstone of vibration based monitoring lies in frequency domain analysis, particularly through Fast Fourier Transform (FFT) and spectral analysis techniques [57]. Spectral analysis is a

way to analyze the vibration data based on the Fourier transform [30], [31]. These methods facilitate the decomposition of vibration signals into time or frequency components, aiding in the identification of fault-related frequencies and harmonics [20], [113]. Monitoring the vibration



signatures of different components in the frequency domain helps in identifying potential faults and assessing the overall health of the wind turbine [57]. It enables early detection of faults, and identification of fault frequencies, and provides valuable awareness for effective maintenance strategies, ultimately contributing to the reliability and longevity of wind turbine systems. **Figure 9** shows the Time and Frequency Domain Analysis

process.

Figure 10.

Hilbert-huang

transform process

Time and frequency domain analysis is widely applied in industry to decompose vibration signals in order to enable early detection of abnormal patterns. Fast Fourier Transform (FFT), for example, isolates fault frequencies in turbine bearings and gearboxes, to intervene before faults escalate. General Electric (GE) employs frequency domain analysis in its turbines to isolate vibration frequencies that correlate with specific faults, such as bearing or gearbox damage.

4.3. Time-Frequency Analysis

Time-frequency analysis techniques, such as the wavelet transform and Hilbert-Huang transform (HHT), offer advantages in capturing transient and non-stationary signals [11], [20], [30], [57]. As shown in Figure 10, the Hilbert-Huang transform is the result of empirical mode decomposition (EMD) and Hilbert spectral analysis (HSA) [118]. Other techniques of time-frequency analysis are the Short-Time Fourier Transform, Cepstral Analysis, Time-Frequency Distribution, Wavelet Packet Transform, and Empirical Mode Decomposition that can be established for detecting blade damage and imbalance in wind turbine blades [60], [119]. Empirical mode decomposition and Hilbert-Huang transform are adaptive signal processing techniques that can decompose non-stationary and nonlinear vibration signals into intrinsic mode functions (IMFs), facilitating the analysis of complex vibration patterns associated with turbine faults [120]. A combination of these time-frequency analysis techniques is often employed to ensure a compreh-



ensive understanding of the vibration signals. This multi-faceted approach enhances the ability to detect and diagnose various mechanical issues, contributing to the overall reliability and performance of wind turbine.

Time-frequency analysis, including wavelet and Hilbert-Huang transforms, is especially useful for handling non-stationary signals from wind turbines. They identify complex fault signals like blade imbalances. This addresses performance issues that would otherwise go undetected, especially in variable operational environments. Vestas uses wavelet transform-based time-frequency analysis to monitor the transient signals of wind turbines.

4.4. Envelope Analysis and Demodulation

Envelope analysis and demodulation techniques focus on extracting fault-related modulations from vibration signals [19], [30]. They are essential techniques in the field of vibration based techniques of condition monitoring, particularly in the context of wind turbines, which are subjected to various environmental and operational conditions, making it critical to monitor their health for optimal performance and to prevent unexpected failures [121]. Envelope analysis involves extracting the envelope of a signal, which represents the variation in amplitude over time. It is applied to vibration signals acquired from sensors placed on critical components such as bearings, gearboxes, and generators for identifying low-frequency modulation components associated with faults like gear tooth damage, bearing defects, and other machinery anomalies [64]. The demodulation process is closely related to envelope analysis and involves separating the high-frequency vibration signal from the low-frequency modulation components [113]. As a result, demodulation helps in isolating the specific fault-related information embedded in the signal. This separation allows for a more detailed analysis of the fault characteristics, making it easier to

identify the root cause of the vibration anomaly. Bearing faults often exhibit amplitude modulations related to defects [53], [122]. Figure 11 shows the demodulation flow chart.



Envelope analysis has practical utility in isolating fault-related modulations within vibration signals. It is needed to identify bearing and gearbox defects. That is, it is used to detect amplitude variations that signify wear and address issues before they lead to catastrophic failures, in order to preserve turbine functionality and safety. Companies like SKF apply envelope analysis to monitor bearings in wind turbines.

4.5. Statistical and Pattern Recognition Techniques

Statistical methods and pattern recognition algorithms, including principal component analysis and artificial neural networks, aid in extracting fault signatures and identifying abnormal patterns within vibration data [65], [123]. These algorithms are integral to vibration analysis, mainly in machinery condition monitoring and fault diagnosis. The statistical techniques include descriptive statistics, probability distributions, hypothesis testing, regression analysis, and time series analysis. This provides the distribution, central tendency, and relationships within vibration data. The pattern recognition algorithms facilitate the extraction of meaningful features from vibration signals and the detection of anomalies or faults. They include supervised and unsupervised machine learning (ML) methods, as shown in Figure 12.



In real-world applications, Principal Component Analysis (PCA) statistical methods simplify vibration data for fault detection. However, machine learning algorithms can classify fault types. These techniques enhance condition monitoring systems by providing quick and accurate diagnostics for efficient decision-making and maintenance prioritization. Pattern recognition is applied by IBM's Watson IoT platform to monitor turbine health using algorithms.

4.6. Integration of Multiple Sensor Data

The wind turbines are equipped with various sensors, including accelerometers, strain gauges, temperature sensors, and oil condition sensors. Integrating data from multiple sensors provides a more comprehensive view of the turbine's health and performance, allowing for early detection of faults and abnormalities [31], [123]. Data fusion techniques combine information from different sensors to improve fault diagnosis accuracy and reliability. For example, combining vibration data with temperature measurements can help identify overheating bearings or gearbox components [47]. Multi-sensor integration, as shown in Figure 13, elevates wind turbine reliability, paving the way for sustainable and efficient energy generation.

By combining data from multiple sensors (such as accelerometers and temperature sensors) a holistic view of turbine health is gained. This integration enables more precise fault detection, identifies overheating issues and indicates component degradation, to extend component life. Siemens Wind Power integrates multiple sensor types to monitor the health of turbine components.



4.7. Remote Monitoring and Internet of Things (IoT) Integration

Remote monitoring involves the continuous surveillance of wind turbine components from a centralized location. This approach leverages advanced wireless sensors, communication technologies, and cloud-based platforms to enable real-time data collection, analysis, and decision-making, ultimately enhancing the reliability and performance of wind energy systems [64]. Cloud computing platforms provide scalable storage and processing capabilities. They are used for analyzing large volumes of vibration data collected from multiple wind turbines. Cloud-based analytics tools can perform advanced data analytics tasks, like trend analysis, pattern recognition, and predictive maintenance modelling. They are using machine learning algorithms and big data techniques [124]. Recent advancements include the integration of vibration monitoring into IoT platforms, to enable remote monitoring and real-time data analysis [15], [123]. The integration of IoT technology in condition monitoring of wind turbine takes this approach to a higher level. IoT



enables the collection and transmission of data from various sensors and devices in real-time. This provides a comprehensive overview of turbine health [121], as shown in Figure 14.

IoT integration in real wind turbine monitoring systems facilitates remote diagnostics through continuously transmitting the vibration data to centralized platforms. This setup supports real-time analysis, responds quickly to anomalies, and reduces the need for frequent on-site inspections. Envision Energy incorporates IoT in its wind turbine fleet for remote monitoring

4.8. Prognostics and Health Management (PHM)

Vibration based prognostics and health management aim to predict component degradation and estimate the remaining useful life based on their existing health condition and operational history [47], [125]. These techniques utilize some advanced data analytics, including statistical modelling, machine learning, and physics-based models. They forecast potential failures and prioritize maintenance activities, as shown in Figure 15. By predicting failures before they occur, Prognostics and Health Management systems enable proactive maintenance strategies, minimize downtime, reduce repair costs, and extend the lifespan of critical components [125]. Prognostics takes vibration based monitoring to the next level by not only detecting current issues but also predicting the future health of critical components. This involves employing some advanced analytics, machine learning algorithms, and predictive modelling to assess the degradation of components and estimate their remaining useful life [126].



Mechanical Engineering for Society and Industry, Vol.4 No.3 (2024)

PHM systems predict component degradation and estimate remaining useful life. This is invaluable in operational turbines. The PHM systems enable proactive maintenance planning to reduce downtime and optimize resource allocation. GE Renewable Energy uses PHM systems to estimate the remaining useful life of critical turbine components based on the current vibration and operational data.

4.9. Machine Learning and Artificial Intelligence Integration

Machine learning and artificial intelligence algorithms are increasingly integrated into condition monitoring systems for wind turbines [48]. These algorithms can analyze vast amounts of vibration data collected from sensors installed on the turbine components. Supervised learning algorithms can be trained on labelled data to classify vibration patterns associated with different types of faults, such as bearing wear, gearbox damage, or blade imbalance [109]. Unsupervised learning techniques, such as clustering and anomaly detection, can identify abnormal vibration patterns that may indicate emerging faults or degradation in turbine components. Reinforcement learning algorithms can optimize maintenance strategies by learning from past maintenance actions



and their outcomes, thus improving the efficiency of maintenance operations over time [127]. The human-cyber-physical system (HCPS) concept, shown in Figure 16, is one of the machine learning and artificial intelligence techniques of upcoming wind turbines [128]. This includes an artificial intelligence that directly controls the wind turbine operation in quasi-realtime.

Machine learning and artificial intelligence in condition monitoring analyze extensively the vibration datasets. They detect patterns associated with specific faults. In practice, artificial intelligence algorithms predict component wear and enable pre-emptive repairs. Siemens uses machine learning algorithms in their turbine monitoring systems in order to detect unusual vibration patterns associated with specific faults.

4.10. Condition-Based Maintenance

Condition-based maintenance strategies use real condition monitoring information to optimize maintenance schedules and prioritize maintenance activities based on the actual health condition of wind turbine components [21], [129]. Condition-based maintenance enables predictive and proactive maintenance actions, such as bearing replacements, gearbox overhauls, or blade inspections, before failures occur [130]. By avoiding unnecessary maintenance tasks and focusing



resources on critical components, Condition-based maintenance strategies can reduce maintenance costs, extend equipment lifespans, and maximize turbine uptime and performance. Figure 17 shows the workflow of the Condition-based maintenance strategy.

Condition-based maintenance strategies control real-time data in order to schedule repairs based on actual equipment conditions rather than fixed intervals. This tailored approach reduces unnecessary costly maintenance and, at the same time ensures turbines operate at optimal efficiency with more sustainable energy production. Suzlon Energy implements condition-based maintenance on its wind turbines by monitoring real-time vibration data for guiding maintenance scheduling.

4.11. Eigensystem Realization Algorithm (ERA)

Eigensystem Realization Algorithm is basically a system identification technique. It is employed to extract the modal parameters (such as natural frequencies, mode shapes, and damping ratios) of a system from measured data. It is based on the realization theory of linear

Figure 17. Workflow of condition-based maintenance strategy

Figure 16.

Condition

concept

monitoring with

machine learning and

artificial intelligence

time-invariant systems and relies on the singular value decomposition of the system's response matrix [131]. In the context of condition monitoring of wind turbine, Eigensystem Realization



Algorithm can be applied to the vibration data collected from the sensors that are installed turbine system. on the Eigensystem Realization Algorithm can identify the natural frequencies and mode shapes of the turbine blades, tower, and other components by analyzing the collected vibration signals. Any variations in these average parameters over time can indicate structural damage, fatigue, or other matters affecting the turbine's performance [59]. Figure 18 shows Eigensystem Realization Algorithm process.

ERA is used in industry to monitor the structural integrity through extracting modal parameters from vibration data. In wind turbines, ERA helps to detect shifts in structural dynamics, indicate wear or damage in blades and towers, and finally enable early intervention before significant structural failures occur. The National Renewable Energy Laboratory (NREL) employs ERA to track structural changes in their turbine towers and blades.

4.12. Natural Excitation Technique (NExT) ERA

Natural Excitation Technique (NExT) is another method for a system identification. It aims to approximately estimate the modal parameters of a system without relying on external excitation sources. It utilizes the natural excitation existing in the structure itself, such as wind-induced vibrations or ambient noise, to extract the modal properties [132], [133]. In condition monitoring of wind turbines, Natural Excitation Technique can be particularly useful because it doesn't require additional external forces to be applied to the turbine. This can be challenging and sometimes impractical in operational wind farms. Natural Excitation Technique can still identify the natural frequencies, damping ratios and mode shapes of the turbine system by analyzing the ambient vibrations and noise picked up by the sensors [133]. Changes in these modal parameters can then be indicative of structural changes or damage that may require maintenance or repair [134]. Figure 19 shows the Natural Excitation Technique Analysis used for wind turbines.

The NExT technique controls ambient vibrations, such as wind-induced motion. It extracts modal properties of turbine components without external excitation. In practice, this method allows for continuous monitoring of structural health, as well as supporting preventive maintenance actions based on real-world operating conditions. Companies like Ørsted apply this technique to track changes in modal parameters.



Eigensystem realization algorithm process

4.13. Observer/Kalman Filter Identification (OKID)

The Observer/Kalman filter Identification is basically a timed-domain and data-driven modelling technique for system identification and fault detection in dynamic systems like wind turbines. It uses input-output data to estimate system dynamics and parameters, constructing mathematical models representing turbine behavior [135]. Observer/Kalman filter Identification facilitates model-based fault detection by predicting expected vibration responses and comparing them with measured data. Its adaptability and predictive maintenance capabilities enable it to anticipate and address issues, leading to optimizing maintenance efforts and maximizing turbine reliability. The Observer/Kalman filter Identification process chart is illustrated in Figure 20.

OKID applies model-based fault detection through estimating system dynamics and predicting expected vibration patterns. In industry, this technique identifies deviations from normal behaviour in turbine components. This is needed for maintaining high reliability in wind energy operations. The OKID technique is applied by companies like Goldwind to model turbine dynamics and predict expected vibration responses.

Vibration-based condition monitoring is a widely used technique for assessing the health of machinery and equipment. Table 6 shows the strengths and limitations of these vibration based techniques of condition monitoring.

Table 6.	Technique	Strengths	Limitations	Potential Improvements
gths and vibration hiques of phitoring	Signal Acquisition and Processing	Integrates vibration data with machine learning for fault classification.	Limited adaptability across different turbine models and environmental conditions.	Testing adaptability across diverse turbine models and environments.
	Frequency Domain Analysis	Useful for analyzing environmental effects on turbine performance.	Only captures frequency information, missing transient faults. High computational	Comparisons with time- frequency methods to improve fault detection.
	Time-Frequency Analysis	disturbances, addressing transient faults.	demands, limiting real-time applicability.	methods or enhance real-time capabilities.
	Envelope Analysis and Demodulation	Effective in identifying bearing faults.	Sensitive to noise, affecting reliability.	Combine with noise-robust techniques, like AI filters, for better fault isolation.
	Statistical and Pattern Recognition	Reliable for fault classification in wind turbine pitch systems.	Limited scalability across diverse fault types.	Broader testing and IoT integration for faster data acquisition and response.
	Integration of Multiple Sensor Data	health monitoring using vibration, torque, and electrical signals.	Data fusion challenges, leading to inconsistencies in fault detection.	Develop robust sensor fusion algorithms for diverse signal sources.
	Remote Monitoring and IoT Integration	Enables remote tracking of turbine parameters, like speed and temperature.	Faces data security issues and slow response times.	Integrate with machine learning for predictive maintenance while ensuring data security.
	Prognostics and Health Management (PHM)	Shows potential for long- term reliability and fault prediction.	Data acquisition challenges across various turbine models, limiting adaptability.	Develop PHM frameworks adaptable to real-time use, leveraging machine learning and sensor data.
	Machine Learning and AI Integration	Optimizes turbine control systems, improving power generation.	High data requirements and model reliability concerns.	Standardize training data and explore models with reduced data dependency.
	Condition-Based Maintenance (CBM)	Reduces maintenance costs, especially with support vector machine classification.	Lacks robustness for handling diverse fault types.	Explore hybrid models combining SVMs with other AI techniques for complex faults.
	Eigensystem Realization Algorithm (ERA)	Sensitive parameter monitoring enhances structural health assessment.	Susceptible to false positives due to operational variability.	Refine algorithm for consistency across different environmental conditions.
	Natural Excitation Technique (NExT)	Provides accurate modal identification of structural elements.	Limited to structural faults, less effective for operational issues like blade or gearbox faults.	Expand applicability to detect a wider range of faults, including non-structural issues.
	Observer/Kalman Filter Identification	Effective for dynamic fault detection in turbine systems.	Computationally complex and challenging to implement in real time.	Simplify or optimize an algorithm for better real-time application.

Strengths and limitations of vibration based techniques of condition monitoring



Vibration based techniques of condition monitoring techniques for wind turbines have been extensively researched to improve reliability and reduce maintenance costs. Various methods, with time synchronous averaging, frequency domain analysis and bearing envelope analysis, have been evaluated on full-scale wind turbine drive trains [88]. While traditional techniques like vibration and oil analysis require multiple sensors and complex computations, cost-effective alternatives have emerged. One approach uses generator output power and rotational speed with continuous wavelet transform to detect faults in both fixed and variable-speed turbines [136]. Another study developed new techniques combining empirical mode decomposition, principal component analysis, and continuous wavelet transform with feature intensity level for early fault detection in turbine blades. Recent trends include acoustic emission for detecting incipient failures and optical fibered sensors for mechanical health monitoring of blades. These advancements aim to enhance wind turbine reliability and reduce downtime. For each of these techniques, Table 7 summarizes published articles.

These vibration based techniques of condition monitoring techniques play a very important role in industrial maintenance by detecting faults in machinery before they escalate into costly failures. In this comparative analysis, these techniques are compared considering their performance, advantages, disadvantages, suitability, and computational costs, as shown in Table 8.

5. Progressions in Vibration based Techniques of Condition Monitoring

The field of vibration based techniques of condition monitoring of wind turbines has witnessed significant progress due to advancements in vibration analysis [60]. Ensuring the best performance and safety of wind turbines is needed for renewable energy generation [110], [111]. vibration analysis acts as an essential diagnostic means for monitoring the health and functionality of these complex systems. Over the years, vibration analysis has revolutionized the detection, analysis, and prediction of potential turbine failures, leading to more efficient maintenance strategies and increased reliability [9], [15]. The core principle behind vibration analysis in condition monitoring of wind turbines involves detecting and interpreting mechanical vibrations within the turbine structure [137]. Analyzing these vibrations offers a clear understanding of the overall health condition of turbine components, and enables proactive maintenance to minimize downtime.

5.1. Integration of Advanced Sensor Technologies

Recent advancements in vibration analysis for wind generators include the combination of recent advanced sensor technologies. Traditional vibrational monitoring structures generally depended on primary sensors to capture facts [31], [138]. However, the advent of high-precision accelerometers, stress gauges, and different advanced sensors has drastically better the information collection abilities [19], [123]. These sensors provide better sampling rates and increased sensitivity. This enables the monitoring of multiple factors concurrently and taking into consideration extra accurate and complete assessments of turbine vibrations. Additionally, the implementation of new wireless sensor networks has streamlined facts collection tactics. It enabled real-time monitoring of turbine vibrations and facilitated non-stop statistics transmission to centralized monitoring systems [139]. These improvements allow for prompt detection of abnormalities, fashion evaluation, and statistics-pushed decision-making with none constraints of bodily wiring. These enhancements encompass the development of state-of-the-art sensors and

data acquisition structures, which give excessive-resolution vibration records [140]. Such advancements have enabled greater unique fault detection and analysis in machinery, as well as decreasing downtime and upkeep costs extensively.

Table 7.	Ref.	Techniques used	Brief Description/Topics	Critical Analysis
Description and critical	[141]	Signal	A method was presented for the	Establishes an initial integration of vibration data and
analysis of published		Acquisition and	health monitoring of wind turbine	machine learning, focusing on blade health. However, lack of
articles about vibration		Processing	blades using vibration signal analysis	exploration of performance across different turbine models
based techniques of			and machine learning techniques for	and environmental conditions, indicating a gap in testing the
condition monitoring			fault classification.	adaptability of this approach.
tochniquos	[142]	Frequency	Frequency-domain analysis of wind	Useful for analyzing environmental effects on turbine
techniques		Domain Analysis	turbine performance in complex	performance, but is limited by its frequency-only focus.
			terrain was used to examine power	Flaws arise in comparing results with time-frequency
			spectral density and frequency	methods, as the latter may capture transient faults that
	[1/2]		Time frequency analysis was used to	Highlights advantages over frequency only methods by
	[143]	Δnalvsis	quantify and compare the time-	canturing dynamic disturbances. However, time-frequency
		Analysis	varving power quality disturbances	methods require high computational power, an identified
			introduced by different wind turbine	gap for real-time applications. Further comparison with
			generator types.	machine learning techniques could enhance predictive
			o <i>n</i>	accuracy.
	[144]	Envelope	Narrowband envelope analysis can	While highly effective for bearing fault detection, envelope
		Analysis and	effectively identify bearing faults in	analysis is sensitive to noise, limiting its reliability. Contrary
		Demodulation	direct-driven wind turbines.	findings on noise interference suggest a gap in combining
				this with noise-robust techniques, such as AI filters, for
				improved fault isolation.
	[145]	Statistical and	A pattern recognition method was	Demonstrates reliability in fault classification but lacks
		Pattern	proposed for fault analysis of wind	scalability across diverse fault types, indicating a need for
		Recognition	turbine pitch systems to improve	broader testing. Combining pattern recognition with real-
		rechniques	reliability.	time for could enhance data acquisition and fault resolution
	[1/6]	Integration of	The use of multi-sensor data was	Specu. Highlights the benefits of multi-modal data but suffers from
	[140]	Multiple Sensor	explored to monitor the health	data fusion challenges. Contradictory findings show
		Data	condition of wind turbine systems.	inconsistencies in fault detection accuracy when integrating
			including vibration, torque, and	sensor types, indicating a gap in robust sensor fusion
			electrical signals,	algorithms for diverse signal sources.
	[147]	Remote	A system was described for remote	Useful for remote fault tracking but limited by data security
		Monitoring and	monitoring and control of wind	and response times. Integrating machine learning with IoT
		IoT Integration	turbine parameters like speed,	for predictive maintenance could address gaps in real-time
			vibration, and temperature using	decision-making while maintaining data security.
			IoT.	
	[148]	Prognostics and	The challenges and opportunities in	PHM shows potential for long-term reliability, though
		Health	applying prognostics and health	challenges in data acquisition from diverse turbine models
		(DUM)	management (PHM) to wind turbine	frameworks adaptable to real time applications loveraging
		(FIIIVI)	components were discussed	machine learning and sensor data integration
	[149]	Machine	A comprehensive review of how Al	Illustrates the potential of AI for optimizing control systems
	[145]	Learning and Al	and machine learning algorithms	but raises concerns about data requirements. Contradictory
		Integration	were used to optimize wind turbine	findings on AI model reliability indicate a need for
		0	control systems and improve wind	standardization in training data, as well as lower data-
			power generation.	dependency models.
	[150]	Condition-Based	A condition-based care policy for	SVM application shows promising cost reductions but lacks
		Maintenance	wind turbines using support vector	robustness across diverse fault types. Exploring hybrid SVM
			machine classification can reduce	models combined with other AI approaches may close gaps
			maintenance costs.	in handling complex, multi-type faults within turbines.
	[151]	Eigensystem	The parameter sensitivity of the	ERA's sensitivity enhances structural monitoring but can lead
		Realization	Eigensystem Realization Algorithm	to faise positives. Contradictory findings on accuracy in
		Algorithm (ERA)	(ERA) was examined for estimating	algorithm to bandle opvironmental variability in wind farms
			size wind turbine	algorithm to handle environmental variability in which farms.
	[132]	Natural	An improved Natural Excitation	The improved NExT offers precision in modal identification
	[101]	Excitation	Technique (NExT) method was	but has limited application to non-structural faults. Research
		Technique	presented for more accurate modal	gaps exist in expanding its applicability to detect operational
		(NExT)	identification of structures	faults, such as blade or gearbox issues.
			compared to the traditional NExT	
			method.	
	[62]	Observer/	The Kalman filter and Luenberger	Effective in dynamic fault detection, but observer techniques
		Kalman Filter	estimator-based observers were	are complex and computationally intensive. Contradictory
		Identification	presented for fault diagnosis in the	results on real-time application feasibility indicate a need for
		(OKID)	pitch system of a wind turbine.	algorithm simplification or optimization for practical use.

Table o.
Comparison of
vibration based
techniques of condition
monitoring

rechnique	Performance	Advantages	Disadvantages	Suitability	Costs
Signal Acquisition	Direct measurement	Direct insight into	Limited to basic	Simple systems with	Low to
and Processing	of vibration signals	machine behaviour	fault detection	known faults	moderate
Frequency Domain	Analysis of frequency	Good for identifying	Limited ability to	Systems with known	Moderate
Analysis	components in	specific frequencies.	detect transient	fault frequencies	
	signals		faults		
Time-Frequency	Captures time-	Suitable for non-	Computationally	Systems with	Moderate to
Analysis	varying spectral	stationary signals	intensive	dynamic operating	high
	content.			conditions	
Envelope Analysis	Focuses on	Effective for	Sensitive to noise	Systems with clear	Moderate
and Demodulation	amplitude variations	detecting amplitude	and signal	fault signatures	
	over time	modulation	distortions		
Statistical and	Utilizes statistical	Robust to noise and	Requires extensive	Systems with diverse	Moderate to
Pattern Recognition	models for fault	signal variations	training data	fault patterns	high
Techniques	detection				
Integration of	Fusion of data from	Enhances fault	Complexity in	Complex systems	Moderate to
Multiple Sensor Data	various sensors	detection accuracy	sensor	with multiple failure	high
			synchronization	modes	
Remote Monitoring	Real-time monitoring	Enables proactive	Dependency on	Distributed systems	Moderate
and IoT Integration	from remote	maintenance	network	or inaccessible	
Due en estise en d	locations	strategies	connectivity		
Prognostics and	Predicts the	Allows for predictive	Relies on accurate	Critical systems	Noderate to
	remaining userul life	maintenance	degradation	requiring high uptime	nign
(PIIVI) Machina Loarning	Utilizes algorithms to	Adaptability to	Comployity in	Suctome with	High
and Al Integration	loarn from data	changing operating	model	complex fault	піgн
and Ar integration	leann nonn uata	conditions	interpretation	nattorns	
Condition-Based	Maintenance	Cost-offective	Requires accurate	Systems with	Low to
Maintenance	triggered by	compared to time-	condition	nredictable failure	moderate
maintenance	equinment	hased methods	monitoring	modes	moderate
	conditions	based methods	monitoring	modes	
Eigensystem	Accurate estimation	Accurate with noisv	Requires high-	Systems with	Moderate to
Realization Algorithm	of modal parameters	data, handles both	quality data,	controlled testing	high
(ERA)		deterministic and	sensitive to model	and high accuracy	0
. ,		stochastic inputs	assumptions	с ,	
Natural Excitation	Provide less precise	Doesn't require	Relies on ambient	Systems with	Low to
Technique (NExT)	estimates	external excitation	excitation, less	controlled excitation	Moderate
		sources, suitable for	accurate than	is impractical	
		operational	controlled		
		monitoring	excitation methods		
Observer/Kalman	System identification	Effective for	Requires accurate	Systems with well-	Moderate to
Filter Identification	technique for fault	identifying system	modelling of the	defined dynamic	high
(OKID)	detection	dynamics	system	models	

5.2. Innovations in Signal Processing and Digital Twins

Innovation in vibration analysis for condition monitoring of wind turbines has been driven by means of the requirement for more correct and early detection of faults. Advanced signal processing methods, like wavelet transforms and experimental mode decomposition [140], were used to improve the capacity to identify and isolate faults in complicated machinery [43], [48], [49]. Furthermore, using virtual twins has grown to be awesome in vibration evaluation for wind turbines. Digital twins are virtual imitations of bodily belongings that simulate their conduct in actual time [152]. By combining sensor information with computational models, those virtual twins can mirror the real turbine overall performance, and permit non-stop monitoring, overall performance optimization, and scenario trying out without disrupting operational turbines. This technology helps a deeper knowledge of vibration dynamics and facilitates predictive upkeep strategies [18]. Additionally, advancements in signal processing techniques have contributed to the more specific and green analysis of vibration statistics. They are extracting valuable information from complicated vibration alerts and enhancing fault detection and prognosis accuracy [43], [48], [49].

5.3. Machine Learning and AI in Predictive Maintenance

Machine learning (ML) and artificial intelligence (AI) enable predictive upkeep via analyzing large quantities of vibration statistics. They pick out patterns and are expecting ability faults before they arise [48], [109]. AI algorithms can understand oblique adjustments in vibration styles, and correlate them with regarded failure modes, after which generate early warnings or protection pointers. This proactive method minimizes downtime and optimizes turbine performance [16]. The integration of ML and AI has caused the development of autonomous vibration based totally

techniques of condition monitoring structures capable of real-time condition monitoring. The use of deep learning knowledge of models became confirmed in automatically diagnosing faults with nominal human involvement [153]. The application of system machine learning algorithms, which include neural networks and guide vector machines, becomes highlighted in studying vibration facts to predict machinery health [72], [76], [153]. These algorithms may be learnt from historical information, to pick out styles and anomalies which could suggest imminent disasters. Hybrid models combine multiple gadgets gaining knowledge of strategies. Neural networks and fuzzy logic hybrid systems showed enhanced fault diagnosis capabilities [153]. These advancements have collectively enhanced the precision, efficiency, and reliability of condition monitoring systems, as well as ensuring the sustained health and performance of critical machinery across industries. Recent advancements in vibration based techniques of condition monitoring have significantly improved the fault detection and diagnosis in industrial machinery. The integration of machine learning and artificial intelligence has revolutionized these techniques, and enabled more accurate and reliable monitoring systems [154]. Deep learning approaches have revealed promise in analyzing vibration signals for fault diagnosis in rotating equipment [155]. These data-driven methods offer computational efficiency and can handle complex problems, and making them particularly useful for bridge health monitoring [140]. Vibration analysis remains the most effective method for predictive maintenance, as it allows for the detection of developing bearing defects before they cause further damage [53], [156]. The combination of vibration based monitoring techniques with machine learning has led to quick digital transformation in keeping the continuous performance of existing structures [157].

6. Challenges in Vibration based Techniques of Condition Monitoring

Vibration based techniques of condition monitoring is a critical step of ensuring the structural integrity and optimal performance of wind turbines [21], [59]. These massive structures harness wind energy, converting it into electricity, but they also face numerous challenges, particularly in the realm of vibration analysis [138]. The nonstationary, stochastic nature of wind turbine vibrations complicates fault diagnosis and requires advanced analysis techniques like spectrogram and bi-spectrum analyses [76], [158]. Offshore wind turbines present additional difficulties due to their remote locations, making data transfer and maintenance costly and time-consuming [159]. Reliability-centered maintenance strategies, including accurate condition monitoring systems, are needed for improving availability and reducing energy costs [30]. However, existing commercial monitoring systems often rely on techniques from other industries, which may not be fully adapted to wind turbine specifics. To address these challenges, researchers are exploring innovative approaches, such as integrating event-based alerts with SCADA data for more effective condition monitoring [159], [160]. As the wind energy sector grows, developing reliable and cost-effective monitoring techniques remains a priority [11]. Researchers worked on understanding these challenges in maintaining the functionality and safety of wind turbines [30], [61], [65]. Table 9 shows some of these challenges.

Table 9.	Category	Challenges
Summary of vibration	Environmental factors	Environmental variability and high operational loads
based techniques of	Structural complexity	Complex mechanical structures and nonlinear behavior
condition monitoring	Operational challenges	Limited accessibility, and fault diagnosis and prognosis
challenges	Data and analysis	Data analysis and interpretation, sensor placement and reliability
5.10.16.1800	Cost and resource management	Cost and resource constraints

One of the primary obstacles in vibration analysis for wind turbines is the complex nature of the data. These machines operate in varying environmental conditions, resulting in diverse vibration patterns influenced by wind speed, turbulence, direction, and other atmospheric factors [55]. Analyzing these multifaceted statistics needs for state-of-the-art algorithms and gear capable of discerning regular operational vibrations from those indicating capability faults or damages in the functioning zones of the power curve.

Moreover, wind turbines are subjected to incessant dynamic loading due to the fluctuating wind forces [52]. This continuous stress can lead to fatigue in materials, causing wear and tear that manifest as changes in vibration patterns [117]. Identifying these subtle alterations amidst the background noise of normal operational vibrations is a significant challenge for effective condition

monitoring [64]. Another hurdle in vibration evaluation is the sheer size and top of wind turbines. Offshore wind turbine, particularly, pose logistical challenges for information collection and analysis. Accessibility problems arise, making it difficult to install and maintain vibration monitoring system. Monitoring structures should be long lasting enough to withstand harsh environmental conditions whilst also being easily serviceable in faraway locations [58], [138]. Furthermore, the structural complexity of wind turbines adds some other layer of issue [22]. These machines comprise numerous components, together with blades, gearboxes, turbines, and bearings, every with its vibration signature [23]. Distinguishing among vibrations originating from different components and determining the foundation reason of abnormalities requires advanced analytical strategies and understanding [43]. Additionally, the presence of fake alarms or ambiguous vibration signatures can impede accurate diagnostics. Factors consisting of adjustments in wind situations or operational modes can result in transient vibrations that may mimic fault signs. Distinguishing between those brief vibrations and actual faults may be very important to save you useless downtime and renovation charges [65]. Integration and standardization of monitoring systems pose any other project in the area of vibration primarily based strategies of situation monitoring [161]. Wind turbine manufacturers regularly use extraordinary monitoring technology and protocols, ensuing in compatibility issues and making information interpretation and contrast throughout specific generators a frightening project. Addressing these challenges is needed to ensure the longevity, reliability, and safety of wind turbine in generating easy and sustainable power [30], [61].

To address the challenges in vibration-based condition monitoring for wind turbines, researchers and industry practitioners have been exploring several potential solutions and innovative approaches. Table 10 shows these proposed solutions and approaches:

Table 10.	Challenge Category	Proposed Solutions/Approaches
Potential solutions and		Using wavelet transform, empirical mode decomposition (EMD), and
approaches to address	Complex Vibration Data	variational mode decomposition (VMD) to separate fault-induced
the challenges in		vibrations from operational signals [140].
vibration-based		Implementing wireless accelerometers and fiber optic sensors for more
condition monitoring	Sensor Technology & Accessibility	accurate data capture [64].
for wind turbines	Sensor reenhology & Accessionity	Positioning sensors on specific turbine components to isolate fault signals
		[49].
		Leveraging machine learning algorithms like deep learning and neural
	Data Interpretation & False Alarms	networks to analyze large datasets, reduce false alarms, and improve
		diagnostic accuracy [71].
		Utilizing 5G and edge computing to enable faster data transfer and real-
	Data Transfer for Offshore Turbines	time processing for remote monitoring and efficient response in offshore
		wind turbines [64].
	Maintenance and Resource	Applying predictive models to estimate component lifespan, optimize
	Management	maintenance schedules, reduce downtime, and extend the operational
	Management	life of turbine parts [92].
		Developing unified data formats and diagnostic criteria for better cross-
	Standardization & System Integration	comparison and benchmarking across turbines from different
		manufacturers [91].

7. Trends in Vibration based Techniques of Condition Monitoring

Vibration primarily based techniques of condition monitoring of wind turbines have visible sizeable advancements in latest years, driven with the aid of the increasing want for reliable and green renewable energy resources. Modern trends on this area emphasized the combination of new advanced sensor technology, like wireless accelerometers and MEMS sensors, which provide better sensitivity and actual-time statistics acquisition. Additionally, the implementation of machine learning algorithms and synthetic intelligence has revolutionized the evaluation of vibration facts, allowing more correct fault detection and predictive upkeep. These technologies facilitate the early identification of issues consisting of imbalance, misalignment, and bearing wear, thereby lowering downtime and maintenance prices. The flow toward cloud-based monitoring structures additionally allows for the centralized evaluation of statistics from more than one turbine, improving the capacity to carry out fleet-extensive health tests [162]. Those developments make contributions to the improved reliability, performance, and lifespan of wind turbine operations.

7.1. Role of Big Data and IoT

Big Data techniques are increasingly essential in the condition monitoring of wind turbines. They address the challenges of managing the massive volumes of data generated by modern sensor technologies [163], [164]. With data output reaching around 2 TB monthly, these Big Data techniques enable efficient processing and analysis of vast datasets. They facilitate the detection of system deterioration and optimize maintenance scheduling [164]. The real-time monitoring and online analysis of wind turbine conditions are possible by combining Big Data with advanced sensors, such as Macro Fiber Composites. Both enhance the ability to track equipment health continuously [165].

Recent advancements extend beyond Big Data to include machine learning and data mining strategies. They are transforming predictive maintenance and health assessment for both individual wind turbines and larger wind farms [38]. These smart analytics, in combination with Big Data processing, significantly improve fault diagnostics, provide early failure predictions, and enhance overall system reliability and economic performance [124]. As condition monitoring technologies progress, the AI based algorithms are becoming essential for processing the complex datasets generated by turbine sensors and refining the accuracy and efficiency of predictive models [38]. The integration of advanced sensor technology has substantially enhanced data collection capabilities within vibration based monitoring of wind turbines [19], [31], [64], [123]. IoT and cloud-based systems are revolutionizing data collection, processing, and remote access. They enable real-time monitoring, predictive analytics, and proactive maintenance planning from virtually any location [166]. This shift allows to rely on early-warning systems that reduce the need for on-site inspections [121].

The AI and IoT technologies are expected to drive the evolution of autonomous, real-time monitoring systems. AI algorithms will further optimize predictive maintenance models by offering more precise forecasts of potential faults, based on both historical and live vibration data [17], [18]. These algorithms are particularly those that continuously learn from incoming data. Additionally, advancements in IoT enabled sensors will continue to improve the depth and quality of data collected and allow for more granular insights into turbine health and performance. As these tools advance, the industry-wide anticipation will shift significantly towards condition-based maintenance strategies [16], [167], [168].

7.2. Potential Impact of New Technologies on Condition Monitoring

Recent advancements in condition monitoring of wind turbine have focused on various technologies and approaches. Vibration analysis, oil debris analysis, and acoustic emission are established condition monitoring techniques for detecting failures in components like gearboxes and bearings [23], [169]. SCADA data has gained attention for cost-effective condition monitoring, with approaches including trending, clustering, and normal behavior modeling [34], [50], [160]. Initial experiences with condition monitoring systems have revealed challenges such as the 3/revolution effect due to tower shadow and wind shear, leading to the development of enhanced time synchronous average algorithms [59], [170]. Emerging technologies for structural health monitoring include optical fiber sensors embedded in blades and radiography inspection methods [22], [169]. Both signal-based and model-based approaches are being explored for fault diagnosis and lifetime prognosis, with ongoing research focusing on improving the reliability and availability of wind turbines [39], [60].

New technologies are revolutionizing the field of vibration based techniques of condition monitoring for wind turbines, leading to increased reliability, reduced downtime, and enhanced performance in the renewable energy sector [9], [171]. The refinement of signal processing algorithms, frequency analysis, time-domain analysis, and spectral analysis techniques provides a deeper understanding of the complex vibration patterns in wind turbines [31], [172]. The implementation of digital twins enables simulation and predictive analysis based on real-time vibration data, aiding in optimizing performance and predicting potential failures [127], [173]. These technological advancements contribute meaningfully to the efficiency and effectiveness of condition monitoring, as well as ensuring the longevity and reliability of wind turbines [152].

7.3. Future Directions

Condition monitoring of wind turbines has become increasingly important as the wind energy sector keeps growing, with a focus on reducing operation and maintenance costs while improving reliability [11]. Current condition monitoring of wind turbine techniques includes vibration analysis

and acoustic emission detection [169]. Time-frequency analysis tools like wavelets are widely used due to the variable-speed nature of wind turbines [169]. Emerging trends in condition monitoring of wind turbine include optical fiber sensors for structural health monitoring, radiography inspection, and Wi-Fi networks for cost-effective communication [59], [169]. The industry is moving towards more advanced turbine designs and reliable, cost-efficient monitoring techniques to enhance availability and reduce maintenance costs [174], [175]. Future challenges involve addressing weaknesses in current condition monitoring of wind turbine practices and defining research priorities to meet the evolving technological and market demands of the wind industry [11], [175].

The current state of vibration based techniques of condition monitoring for wind turbines highlights notable technological advancements and identifies several critical gaps requiring further research and development [15]. Recent studies have aimed at refining signal processing techniques, integrating machine learning algorithms, and enhancing remote monitoring capabilities in vibration based techniques of condition monitoring for wind turbines. Specifically, advanced signal processing methods like envelope analysis and demodulation have proven essential in bearing fault detection [45], [92]. Moreover, the role of machine learning in fault classification and estimating remaining useful life is becoming increasingly prominent. However, significant challenges persist, such as the need for improved fault diagnostics and severity assessments [19]. Standardization of diagnostic protocols and thresholds remains inconsistent [91]. The computational complexity, especially in remote monitoring, is a considerable challenge [58], [64]. Future research should focus on developing enhanced methodologies for fault severity assessment, as well as establishing standardized diagnostic protocols, advancing real-time analytics and edge computing, and integrating robust multi-sensor fusion techniques [15], [43], [44], [61]. Large-scale field tests and validation studies are also important to verify the efficacy and reliability of these techniques under different operating conditions and wind turbine types [19].

8. Conclusions and Recommendations

8.1. Conclusions

The literature overview on vibration based techniques of condition monitoring of wind turbines well-known shows that this approach is both powerful and important to make sure the reliability and performance of wind strength structures. Vibration based monitoring allows early detection of mechanical faults, together with bearing wear, gearbox issues, and blade imbalances, thereby mitigating the risk of unexpected failures and decreasing renovation costs [68]. The potential to display actual-time vibrations and analyze patterns allows for predictive renovation techniques, which might be critical in optimizing turbine performance and lengthening operational lifespans.

Comparatively, vibration based totally monitoring stands proud in opposition to other situation monitoring strategies inclusive of oil analysis, thermal imaging and acoustic emission. While every technique has its merits, vibration based totally monitoring offers an extra direct and quantifiable approach for detecting mechanical anomalies. It presents continuous, real-time records which can be analyzed to are expecting any capability disasters, not like other techniques which can be greater situational or much less complete in their diagnostics.

The important analysis of the reviewed techniques highlights several trends. Developments in sensor generation, data acquisition, and machine learning algorithms advanced significantly the accuracy and efficiency of vibration primarily based monitoring structures. However, challenges continue to be in integrating those systems with current turbine infrastructure and coping with the bigger volumes of generated information. The variability in turbine designs and operational situations also complicates the standardization of vibration monitoring practices.

8.2. Recommendations for Future Research:

- Integration with Predictive Analytics: Future research needs to focus on enhancing the integration of vibration data with advanced predictive analytics and machine learning models. This will improve fault detection accuracy and enable more precise forecasting of component failures.
- Standardization and Benchmarking: Developing standardized protocols and benchmarks for vibration monitoring across different turbine models and manufacturers will facilitate better comparison and validation of monitoring systems and techniques.

- Sensor Technology Advancements: Continued research into the development of more durable, cost-effective, and high-resolution sensors will further enhance the capabilities of vibration based monitoring systems.
- Data Management and Interpretation: Addressing the challenges associated with managing and interpreting large volumes of vibration data is important. Research should explore efficient data processing methods and algorithms to extract actionable insights and reduce data overload.
- Integration of AI and IoT for real-time for autonomous fault detection in wind turbines and focusing on accurate predictive models that adapt to varying conditions. Efforts should consider data standardization, interoperability, and cybersecurity to support robust and scalable implementation.
- Integration with Other Monitoring Techniques: Combining vibration based monitoring with other condition monitoring techniques, such as acoustic emission and thermal imaging, could provide a more comprehensive approach to condition assessment and failure prediction.
- Field Studies and Real-world Validation: Conducting field studies and real-world validation of advanced vibration monitoring systems will help assess their practical effectiveness and reliability in diverse operational environments.

The future research can further enhance the effectiveness of vibration based condition monitoring by addressing these areas and contributing to the advancement of wind turbine technology and the broader wind energy industry.

Acknowledgements

The authors also acknowledge the contributions of the individual authors cited and referenced in this paper, whose efforts were outstanding.

Authors' Declaration

Authors' contributions and responsibilities - A. H. A. designed the framework of this paper and was a major contributor in writing the manuscript. K. C. V. was a major contributor in sections 2, 3 and 4. V. V. K. was also a major contributor in revising the manuscript and improving sections 5, 6, 7 and 8. All authors read and approved the final manuscript.

Funding – The article publication cost is covered by the authors.

Availability of data and materials - The data underlying this study are derived from publicly available sources, including published journals and reputable online repositories, as cited within the text.

Competing interests - The authors declare no competing interest.

Additional information – No additional information from the authors.

References

- O. S. Samiani and M. Boroushaki, "Optimal design of Archimedes Wind Turbine using genetic algorithm," vol. 314, no. November 2024, 2025.
- [2] E. Achdi, B. F. T. Kiono, S. H. Winoto, and M. Facta, "Improving cross-axis wind turbine performance: A Lab-scale investigation of rotor size and blades number," *Mechanical Engineering for Society and Industry*, vol. 4, no. 1, pp. 82–91, Jul. 2024, doi: 10.31603/mesi.10837.
- [3] A. D. Korawan and R. Febritasari, "Experimental investigations of number of blades effect on archimedes spiral wind turbine performance," *Mechanical Engineering for Society and Industry*, vol. 4, no. 2, 2024, doi: 10.31603/mesi.12373.
- [4] S. Roga, S. Bardhan, Y. Kumar, and S. K. Dubey, "Recent technology and challenges of wind energy generation: A review," *Sustainable Energy Technologies and Assessments*, vol. 52, p. 102239, 2022, doi: 10.1016/j.seta.2022.102239.
- [5] Global Wind Energy Council, "Global Wind Report 2023," *Global Wind Energy Council*, 2023.

- [6] U. Singh, M. Rizwan, H. Malik, and F. P. García Márquez, "Wind Energy Scenario, Success and Initiatives towards Renewable Energy in India—A Review," *Energies*, vol. 15, no. 6, p. 2291, 2022, doi: 10.3390/en15062291.
- [7] A. S. Darwish and R. Al-Dabbagh, "Wind energy state of the art: present and future technology advancements," *Renewable Energy and Environmental Sustainability*, vol. 5, p. 7, 2020, doi: 10.1051/rees/2020003.
- [8] P. Sadorsky, "Wind energy for sustainable development: Driving factors and future outlook," *Journal of Cleaner Production*, vol. 289, p. 125779, 2021, doi: 10.1016/j.jclepro.2020.125779.
- [9] T. W. Verbruggen, "Wind Turbine Operation and Maintenance based on Condition Monitoring WT-{omega}," Petten (Netherlands), 2003.
- [10] W. Qiao and D. Lu, "A Survey on Wind Turbine Condition Monitoring and Fault Diagnosis— Part I: Components and Subsystems," *IEEE Transactions on Industrial Electronics*, vol. 62, no. 10, pp. 6536–6545, 2015, doi: 10.1109/tie.2015.2422112.
- [11] P. Tchakoua, R. Wamkeue, M. Ouhrouche, F. Slaoui-Hasnaoui, T. Tameghe, and G. Ekemb, "Wind Turbine Condition Monitoring: State-of-the-Art Review, New Trends, and Future Challenges," *Energies*, vol. 7, no. 4, pp. 2595–2630, 2014, doi: 10.3390/en7042595.
- [12] X.-Y. He, T.-W. Zhao, H.-N. Li, and J. Zhang, "Multi-dimensional seismic response control of offshore platform structures with viscoelastic dampers (II-Experimental study)," *Structural Monitoring and Maintenance*, vol. 3, no. 2, pp. 175–194, 2016, doi: 10.12989/smm.2016.3.2.175.
- [13] P. Bórawski, A. Bełdycka-Bórawska, K. J. Jankowski, B. Dubis, and J. W. Dunn, "Development of wind energy market in the European Union," *Renewable Energy*, vol. 161, pp. 691–700, 2020, doi: 10.1016/j.renene.2020.07.081.
- [14] W. Musial, "Offshore wind market report: 2023 edition," Golden, CO, USA, 2023.
- [15] C. Choe Wei Chang *et al.*, "Recent advancements in condition monitoring systems for wind turbines: A review," *Energy Reports*, vol. 9, pp. 22–27, 2023, doi: 10.1016/j.egyr.2023.08.061.
- [16] W. Y. Liu, B. P. Tang, J. G. Han, X. N. Lu, N. N. Hu, and Z. Z. He, "The structure healthy condition monitoring and fault diagnosis methods in wind turbines: A review," *Renewable and Sustainable Energy Reviews*, vol. 44, pp. 466–472, 2015, doi: 10.1016/j.rser.2014.12.005.
- [17] L. Alhmoud and B. Wang, "A review of the state-of-the-art in wind-energy reliability analysis," *Renewable and Sustainable Energy Reviews*, vol. 81, pp. 1643–1651, 2018, doi: 10.1016/j.rser.2017.05.252.
- [18] Interplay Learning, "Different Types of Maintenance," Interplay Learning, 2020. .
- [19] R. Orsagh, H. Lee, and M. Watson, "Advanced vibration monitoring for wind turbine health management," *Impact Technologies*, no. April, 2006.
- [20] P. Zhang and D. Lu, "A Survey of Condition Monitoring and Fault Diagnosis toward Integrated O&M for Wind Turbines," *Energies*, vol. 12, no. 14, p. 2801, 2019, doi: 10.3390/en12142801.
- [21] R. Yan, S. Dunnett, and L. Jackson, "Impact of condition monitoring on the maintenance and economic viability of offshore wind turbines," *Reliability Engineering & Computer System Safety*, vol. 238, p. 109475, 2023, doi: 10.1016/j.ress.2023.109475.
- [22] R. Valdez-Yepez, C. Tutivén, and Y. Vidal, "Structural health monitoring of jacket-type support structures in offshore wind turbines: A comprehensive dataset for bolt loosening detection through vibrational analysis," *Data in brief*, vol. 53, p. 110222, Feb. 2024, doi: 10.1016/j.dib.2024.110222.
- [23] Y. Kong, Q. Han, F. Chu, Y. Qin, and M. Dong, "Spectral ensemble sparse representation classification approach for super-robust health diagnostics of wind turbine planetary gearbox," *Renewable Energy*, vol. 219, p. 119373, 2023, doi: 10.1016/j.renene.2023.119373.
- [24] Z. Daneshi-Far, G. A. Capolino, and H. Henao, "Review of failures and condition monitoring in wind turbine generators," *The XIX International Conference on Electrical Machines - ICEM* 2010. IEEE, pp. 1–6, 2010, doi: 10.1109/icelmach.2010.5608150.

- [25] S. Sheng and P. Veers, "Wind turbine drivetrain condition monitoring An overview," in Mechanical Failures Prevention Group: Applied Systems Health Management Conference 2011, 2011, no. October.
- [26] I. El-Thalji and E. Jantunen, "On the Development of Condition Based Maintenance Strategy for Offshore Wind Farm: Requirement Elicitation Process," *Energy Procedia*, vol. 24, pp. 328–339, 2012, doi: 10.1016/j.egypro.2012.06.116.
- [27] M. Lydia and G. Edwin Prem Kumar, "Condition monitoring in wind turbines," in Non-Destructive Testing and Condition Monitoring Techniques in Wind Energy, Elsevier, 2023, pp. 229–247.
- [28] P. Tchakoua, R. Wamkeue, T. A. Tameghe, and G. Ekemb, "A review of concepts and methods for wind turbines condition monitoring," 2013 World Congress on Computer and Information Technology (WCCIT). IEEE, pp. 1–9, 2013, doi: 10.1109/wccit.2013.6618706.
- [29] M. C. M. Gowda, N. P. Mallikarjun, P. Gowda, and R. Chandrashekhar, "Improvement of the performance of Wind Turbine Generator using Condition Monitoring techniques," 2013 7th International Conference on Intelligent Systems and Control (ISCO). IEEE, pp. 495–501, 2013, doi: 10.1109/isco.2013.6481205.
- [30] W. Yang, P. J. Tavner, C. J. Crabtree, Y. Feng, and Y. Qiu, "Wind turbine condition monitoring: technical and commercial challenges," *Wind Energy*, vol. 17, no. 5, pp. 673–693, 2012, doi: 10.1002/we.1508.
- [31] A. A. Salem, A. Abu-Siada, and S. Islam, "Condition Monitoring Techniques of the Wind Turbines Gearbox and Rotor," *International Journal of Electrical Energy*, pp. 53–56, 2014, doi: 10.12720/ijoee.2.1.53-56.
- [32] P. Zhang and Z. Chen, "Non-invasive condition monitoring and diagnostics techniques for wind turbines," 2016 IEEE 8th International Power Electronics and Motion Control Conference (IPEMC-ECCE Asia). IEEE, pp. 3249–3254, 2016, doi: 10.1109/ipemc.2016.7512815.
- [33] P. Bangalore, S. Letzgus, D. Karlsson, and M. Patriksson, "An artificial neural network-based condition monitoring method for wind turbines, with application to the monitoring of the gearbox," *Wind Energy*, vol. 20, no. 8, pp. 1421–1438, 2017, doi: 10.1002/we.2102.
- [34] J. Tautz-Weinert and S. J. Watson, "Using SCADA data for wind turbine condition monitoring

 a review," *IET Renewable Power Generation*, vol. 11, no. 4, pp. 382–394, 2016, doi: 10.1049/iet-rpg.2016.0248.
- [35] M. L. Hossain, A. Abu-Siada, and S. M. Muyeen, "Methods for Advanced Wind Turbine Condition Monitoring and Early Diagnosis: A Literature Review," *Energies*, vol. 11, no. 5, p. 1309, 2018, doi: 10.3390/en11051309.
- [36] E. Hoxha, Y. Vidal, and F. Pozo, "Supervised classification with scada data for condition monitoring of wind turbines," in *9th ECCOMAS Thematic Conference on Smart Structures and Materials, SMART 2019*, 2019, pp. 263–274.
- [37] J. Maldonado-Correa, S. Martín-Martínez, E. Artigao, and E. Gómez-Lázaro, "Using SCADA Data for Wind Turbine Condition Monitoring: A Systematic Literature Review," *Energies*, vol. 13, no. 12, p. 3132, 2020, doi: 10.3390/en13123132.
- [38] M. Benbouzid, T. Berghout, N. Sarma, S. Djurović, Y. Wu, and X. Ma, "Intelligent Condition Monitoring of Wind Power Systems: State of the Art Review," *Energies*, vol. 14, no. 18, p. 5967, 2021, doi: 10.3390/en14185967.
- [39] H. Badihi, Y. Zhang, B. Jiang, P. Pillay, and S. Rakheja, "A Comprehensive Review on Signal-Based and Model-Based Condition Monitoring of Wind Turbines: Fault Diagnosis and Lifetime Prognosis," *Proceedings of the IEEE*, vol. 110, no. 6, pp. 754–806, 2022, doi: 10.1109/jproc.2022.3171691.
- [40] R. K. Pandit, D. Astolfi, and I. Durazo Cardenas, "A Review of Predictive Techniques Used to Support Decision Making for Maintenance Operations of Wind Turbines," *Energies*, vol. 16, no. 4, p. 1654, 2023, doi: 10.3390/en16041654.
- [41] F. Castellani, F. Natili, D. Astolfi, and Y. Vidal, "Wind turbine gearbox condition monitoring through the sequential analysis of industrial SCADA and vibration data," *Energy Reports*, vol. 12, pp. 750–761, 2024, doi: 10.1016/j.egyr.2024.06.041.
- [42] S. K. Sahu, V. Kumar, S. Chandra Dutta, R. Sarkar, S. Bhattacharya, and P. Debnath,

"Structural safety of offshore wind turbines: Present state of knowledge and future challenges," *Ocean Engineering*, vol. 309, p. 118383, 2024, doi: 10.1016/j.oceaneng.2024.118383.

- [43] A. Nagy and I. Jahn, "Advanced Data Acquisition System for Wind Energy Applications," *Periodica Polytechnica Transportation Engineering*, vol. 47, no. 2, pp. 124–130, 2018, doi: 10.3311/pptr.11515.
- [44] A. Günel, T. Bley, A. Meshram, M. Klusch, and A. Schütze, "D8.1 Statistical and Semantic Multisensor Data Evaluation for Fluid Condition Monitoring in Wind Turbines," *Proceedings SENSOR 2013*. AMA Service GmbH, Von-Münchhausen-Str. 49, 31515 Wunstorf, Germany, pp. 604–609, 2013, doi: 10.5162/sensor2013/d8.1.
- [45] S. Nese, O. Kilic, and T. C. Akinci, "Condition monitoring with signal processing in wind turbines," *Journal of Vibroengineering*, vol. 13, p. 439, Sep. 2011.
- [46] Z. Zhang and A. Kusiak, "Monitoring Wind Turbine Vibration Based on SCADA Data," *Journal of Solar Energy Engineering*, vol. 134, no. 2, 2012, doi: 10.1115/1.4005753.
- [47] R. Dupuis, "Application of Oil Debris Monitoring For Wind Turbine Gearbox Prognostics and Health Management," Annual Conference of the PHM Society, vol. 2, no. 1, 2010, doi: 10.36001/phmconf.2010.v2i1.1867.
- [48] R. L. Hu, K. Leahy, I. C. Konstantakopoulos, D. M. Auslander, C. J. Spanos, and A. M. Agogino, "Using Domain Knowledge Features for Wind Turbine Diagnostics," 2016 15th IEEE International Conference on Machine Learning and Applications (ICMLA). IEEE, pp. 300–307, 2016, doi: 10.1109/icmla.2016.0056.
- [49] M. Kordestani, M. Rezamand, R. Carriveau, D. S. K. Ting, and M. Saif, "Failure Diagnosis of Wind Turbine Bearing Using Feature Extraction and a Neuro-Fuzzy Inference System (ANFIS)," *Lecture Notes in Computer Science*. Springer International Publishing, pp. 545– 556, 2019, doi: 10.1007/978-3-030-20521-8_45.
- [50] F. Harrou, K. R. Kini, M. Madakyaru, and Y. Sun, "Uncovering sensor faults in wind turbines: An improved multivariate statistical approach for condition monitoring using SCADA data," *Sustainable Energy, Grids and Networks*, vol. 35, p. 101126, 2023, doi: 10.1016/j.segan.2023.101126.
- [51] F. C. Mehlan and A. R. Nejad, "Rotor imbalance detection and diagnosis in floating wind turbines by means of drivetrain condition monitoring," *Renewable Energy*, vol. 212, pp. 70– 81, 2023, doi: 10.1016/j.renene.2023.04.102.
- [52] A. S. Verma, J. Yan, W. Hu, Z. Jiang, W. Shi, and J. J. E. Teuwen, "A review of impact loads on composite wind turbine blades: Impact threats and classification," *Renewable and Sustainable Energy Reviews*, vol. 178, p. 113261, 2023, doi: 10.1016/j.rser.2023.113261.
- [53] Y. Zhang *et al.*, "Integrated intelligent fault diagnosis approach of offshore wind turbine bearing based on information stream fusion and semi-supervised learning," *Expert Systems with Applications*, vol. 232, p. 120854, 2023, doi: 10.1016/j.eswa.2023.120854.
- [54] F. Zhang, M. Chen, Y. Zhu, K. Zhang, and Q. Li, "A Review of Fault Diagnosis, Status Prediction, and Evaluation Technology for Wind Turbines," *Energies*, vol. 16, no. 3, p. 1125, 2023, doi: 10.3390/en16031125.
- [55] M. R. Machado and M. Dutkiewicz, "Wind turbine vibration management: An integrated analysis of existing solutions, products, and Open-source developments," *Energy Reports*, vol. 11, pp. 3756–3791, 2024, doi: 10.1016/j.egyr.2024.03.014.
- [56] C. Velandia-Cardenas, Y. Vidal, and F. Pozo, "Wind Turbine Fault Detection Using Highly Imbalanced Real SCADA Data," *Energies*, vol. 14, no. 6, p. 1728, 2021, doi: 10.3390/en14061728.
- [57] B. Lu, Y. Li, X. Wu, and Z. Yang, "A review of recent advances in wind turbine condition monitoring and fault diagnosis," 2009 IEEE Power Electronics and Machines in Wind Applications. IEEE, pp. 1–7, 2009, doi: 10.1109/pemwa.2009.5208325.
- [58] X. Wu, H. Wang, G. Jiang, P. Xie, and X. Li, "Monitoring Wind Turbine Gearbox with Echo State Network Modeling and Dynamic Threshold Using SCADA Vibration Data," *Energies*, vol. 12, no. 6, p. 982, 2019, doi: 10.3390/en12060982.
- [59] Z.-Q. Xiang *et al.*, "Vibration-based health monitoring of the offshore wind turbine tower using machine learning with Bayesian optimisation," *Ocean Engineering*, vol. 292, p.

116513, 2024, doi: 10.1016/j.oceaneng.2023.116513.

- [60] K. Kong, K. Dyer, C. Payne, I. Hamerton, and P. M. Weaver, "Progress and Trends in Damage Detection Methods, Maintenance, and Data-driven Monitoring of Wind Turbine Blades – A Review," *Renewable Energy Focus*, vol. 44, pp. 390–412, 2023, doi: 10.1016/j.ref.2022.08.005.
- [61] C. J. Crabtree, "Condition Monitoring of Wind Turbines: Challenges and opportunities," in *5th PhD Seminar on Wind Energy in Europe*, 2008, pp. 1–4.
- [62] Z. Zemali, L. Cherroun, N. Hadroug, M. Nadour, and A. Hafaifa, "Fault diagnosis-based observers using Kalman filters and Luenberger estimators: Application to the pitch system fault actuators," *Diagnostyka*, vol. 24, no. 1, pp. 1–13, 2023, doi: 10.29354/diag/161307.
- [63] J. Leng, G. Li, and L. Duan, "The impact of extreme wind conditions and yaw misalignment on the aeroelastic responses of a parked offshore wind turbine," *Ocean Engineering*, vol. 313, p. 119403, 2024, doi: 10.1016/j.oceaneng.2024.119403.
- [64] A. Czyżewski, "Remote Health Monitoring of Wind Turbines Employing Vibroacoustic Transducers and Autoencoders," *Frontiers in Energy Research*, vol. 10, 2022, doi: 10.3389/fenrg.2022.858958.
- [65] B. S. Dhillon, "Corrective Maintenance," in *Engineering Maintenance*, CRC Press, 2002.
- [66] Y. Wang, W. Lu, K. Dai, M. Yuan, and S.-E. Chen, "Dynamic Study of a Rooftop Vertical Axis Wind Turbine Tower Based on an Automated Vibration Data Processing Algorithm," *Energies*, vol. 11, no. 11, p. 3135, 2018, doi: 10.3390/en11113135.
- [67] H. Li, J. Deng, S. Yuan, P. Feng, and D. D. K. Arachchige, "Monitoring and Identifying Wind Turbine Generator Bearing Faults Using Deep Belief Network and EWMA Control Charts," *Frontiers in Energy Research*, vol. 9, 2021, doi: 10.3389/fenrg.2021.799039.
- [68] Z. Gao and X. Liu, "An Overview on Fault Diagnosis, Prognosis and Resilient Control for Wind Turbine Systems," *Processes*, vol. 9, no. 2, p. 300, 2021, doi: 10.3390/pr9020300.
- [69] K. Lin, J. Pan, Y. Xi, Z. Wang, and J. Jiang, "Vibration anomaly detection of wind turbine based on temporal convolutional network and support vector data description," *Engineering Structures*, vol. 306, p. 117848, 2024, doi: 10.1016/j.engstruct.2024.117848.
- [70] X. Gong and W. Qiao, "Imbalance Fault Detection of Direct-Drive Wind Turbines Using Generator Current Signals," *IEEE Transactions on Energy Conversion*, vol. 27, no. 2, pp. 468– 476, 2012, doi: 10.1109/tec.2012.2189008.
- [71] T. Wang, Q. Han, F. Chu, and Z. Feng, "Vibration based condition monitoring and fault diagnosis of wind turbine planetary gearbox: A review," *Mechanical Systems and Signal Processing*, vol. 126, pp. 662–685, 2019, doi: 10.1016/j.ymssp.2019.02.051.
- [72] H. Han and D. Yang, "Correlation analysis based relevant variable selection for wind turbine condition monitoring and fault diagnosis," *Sustainable Energy Technologies and Assessments*, vol. 60, p. 103439, 2023, doi: 10.1016/j.seta.2023.103439.
- [73] P. Santos, L. F. Villa, A. Reñones, A. Bustillo, and J. Maudes, "An SVM-based solution for fault detection in wind turbines," *Sensors (Basel, Switzerland)*, vol. 15, no. 3, pp. 5627–5648, Mar. 2015, doi: 10.3390/s150305627.
- [74] G. Jiang, H. He, J. Yan, and P. Xie, "Multiscale Convolutional Neural Networks for Fault Diagnosis of Wind Turbine Gearbox," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 4, pp. 3196–3207, 2019, doi: 10.1109/tie.2018.2844805.
- [75] L. Xiang, P. Wang, X. Yang, A. Hu, and H. Su, "Fault detection of wind turbine based on SCADA data analysis using CNN and LSTM with attention mechanism," *Measurement*, vol. 175, p. 109094, 2021, doi: 10.1016/j.measurement.2021.109094.
- [76] Q. Yao, H. Bing, G. Zhu, L. Xiang, and A. Hu, "A novel stochastic process diffusion model for wind turbines condition monitoring and fault identification with multi-parameter information fusion," *Mechanical Systems and Signal Processing*, vol. 214, p. 111397, 2024, doi: 10.1016/j.ymssp.2024.111397.
- [77] R. Uma Maheswari and R. Umamaheswari, "Trends in non-stationary signal processing techniques applied to vibration analysis of wind turbine drive train – A contemporary survey," *Mechanical Systems and Signal Processing*, vol. 85, pp. 296–311, 2017, doi: 10.1016/j.ymssp.2016.07.046.
- [78] F. Cheng, L. Qu, and W. Qiao, "Fault Prognosis and Remaining Useful Life Prediction of Wind

Turbine Gearboxes Using Current Signal Analysis," *IEEE Transactions on Sustainable Energy*, vol. 9, no. 1, pp. 157–167, 2018, doi: 10.1109/tste.2017.2719626.

- [79] R. Pandit, D. Astolfi, J. Hong, D. Infield, and M. Santos, "SCADA data for wind turbine datadriven condition/performance monitoring: A review on state-of-art, challenges and future trends," *Wind Engineering*, vol. 47, no. 2, pp. 422–441, 2022, doi: 10.1177/0309524x221124031.
- [80] D. Schlinkert and K. G. van den Boogaart, "The development of the market for rare earth elements: Insights from economic theory," *Resources Policy*, vol. 46, pp. 272–280, 2015, doi: 10.1016/j.resourpol.2015.10.010.
- [81] J. Lei *et al.*, "Techno-Economic Assessment of a Full-Chain Hydrogen Production by Offshore Wind Power," *Energies*, vol. 17, no. 11, p. 2447, 2024, doi: 10.3390/en17112447.
- [82] S. Gasperin and J. Emden, "A second wind: Maximising the economic opportunity for UK wind manufacturing," *Institute for Public Policy Reserch*, 2024. .
- [83] R. B. Randall, Vibration–based Condition Monitoring. Wiley, 2021.
- [84] P. Jayaswal and B. Agrawal, "New trends in Wind Turbine Condition Monitoring System," International Journal of Emerging trends in Engineering and Development, vol. 3, no. 1, pp. 133–148, 2011.
- [85] X. Yuan and L. Cai, "Variable amplitude Fourier series with its application in gearbox diagnosis—Part I: Principle and simulation," *Mechanical Systems and Signal Processing*, vol. 19, no. 5, pp. 1055–1066, 2005, doi: 10.1016/j.ymssp.2004.10.011.
- [86] J. Zhan, J. Huang, L. Niu, X. Peng, D. Deng, and S. Cheng, "Study of the key technologies of electric power big data and its application prospects in smart grid," 2014 IEEE PES Asia-Pacific Power and Energy Engineering Conference (APPEEC). IEEE, pp. 1–4, 2014, doi: 10.1109/appeec.2014.7066162.
- [87] B. K. Bose, "Artificial Intelligence Techniques in Smart Grid and Renewable Energy Systems—Some Example Applications," *Proceedings of the IEEE*, vol. 105, no. 11, pp. 2262– 2273, 2017, doi: 10.1109/jproc.2017.2756596.
- [88] D. Siegel, W. Zhao, E. Lapira, M. AbuAli, and J. Lee, "A comparative study on vibration-based condition monitoring algorithms for wind turbine drive trains," *Wind Energy*, vol. 17, no. 5, pp. 695–714, 2013, doi: 10.1002/we.1585.
- [89] Y. Ou, E. N. Chatzi, V. K. Dertimanis, and M. D. Spiridonakos, "Vibration-based experimental damage detection of a small-scale wind turbine blade," *Structural Health Monitoring*, vol. 16, no. 1, pp. 79–96, 2016, doi: 10.1177/1475921716663876.
- [90] Z. Zhang, A. Verma, and A. Kusiak, "Fault Analysis and Condition Monitoring of the Wind Turbine Gearbox," *IEEE Transactions on Energy Conversion*, vol. 27, no. 2, pp. 526–535, 2012, doi: 10.1109/tec.2012.2189887.
- [91] L. Cao, J. Zhang, Z. Qian, Z. Meng, and J. Li, "Condition monitoring of wind turbine based on a novel spatio-temporal feature aggregation network integrated with adaptive threshold interval," Advanced Engineering Informatics, vol. 62, p. 102676, 2024, doi: 10.1016/j.aei.2024.102676.
- [92] W. Yang, P. J. Tavner, S. Sheng, and R. Court, "Information entropy: An effective approach for wind turbine condition monitoring," in *European Wind Energy Conference and Exhibition* 2012, EWEC 2012, Jan. 2012, pp. 772–778.
- [93] R. Zimroz, J. Urbanek, T. Barszcz, W. Bartelmus, F. Millioz, and N. Martin, "Measurement of Instantaneous Shaft Speed by Advanced Vibration Signal Processing - Application to Wind Turbine Gearbox," *Metrology and Measurement Systems*, vol. 18, no. 4, 2011, doi: 10.2478/v10178-011-0066-4.
- [94] M. V KiranKumar, M. Lokesha, S. Kumar, and A. Kumar, "Review on Condition Monitoring of Bearings using vibration analysis techniques.," *IOP Conference Series: Materials Science and Engineering*, vol. 376, p. 12110, 2018, doi: 10.1088/1757-899x/376/1/012110.
- [95] M. N. Kotzalas and G. L. Doll, "Tribological advancements for reliable wind turbine performance," *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, vol. 368, no. 1929, pp. 4829–4850, 2010, doi: 10.1098/rsta.2010.0194.
- [96] K. Bassett, "Vibration Based Structural Health Monitoring for Utility Scale Wind Turbines,"

Theory of Computing Systems, vol. 20, Jan. 2010.

- [97] A. M. Abdelrhman, M. S. Leong, S. A. M. Saeed, and S. M. A. A.-O. Al Obiadi, "A Review of Vibration Monitoring as a Diagnostic Tool for Turbine Blade Faults," *Applied Mechanics and Materials*, vol. 229–231, pp. 1459–1463, 2012, doi: 10.4028/www.scientific.net/amm.229-231.1459.
- [98] J. Kumar Sasmal, "Condition Monitoring of Wind Turbine Gearbox by Vibration Analysis," International Journal of Scientific Engineering and Research, vol. 3, no. 3, pp. 52–56, 2015, doi: 10.70729/ijser1510.
- [99] A. Romero, Y. Lage, S. Soua, B. Wang, and T.-H. Gan, "Vestas V90-3MW Wind Turbine Gearbox Health Assessment Using a Vibration-Based Condition Monitoring System," *Shock* and Vibration, vol. 2016, pp. 1–18, 2016, doi: 10.1155/2016/6423587.
- [100] A. Joshuva. and V. Sugumaran., "A data driven approach for condition monitoring of wind turbine blade using vibration signals through best-first tree algorithm and functional trees algorithm: A comparative study," *ISA Transactions*, vol. 67, pp. 160–172, 2017, doi: 10.1016/j.isatra.2017.02.002.
- [101] G. Oliveira, F. Magalhães, Á. Cunha, and E. Caetano, "Vibration-based damage detection in a wind turbine using 1 year of data," *Structural Control and Health Monitoring*, vol. 25, no. 11, p. e2238, 2018, doi: 10.1002/stc.2238.
- [102] S. Koukoura, J. Carroll, and A. McDonald, "On the use of AI based vibration condition monitoring of wind turbine gearboxes," *Journal of Physics: Conference Series*, vol. 1222, no. 1, p. 12045, 2019, doi: 10.1088/1742-6596/1222/1/012045.
- [103] D. Saputra and K. Marhadi, "Automatic Fault Diagnosis in Wind Turbine Condition Monitoring," *PHM Society European Conference*, vol. 5, no. 1, p. 8, 2020, doi: 10.36001/phme.2020.v5i1.1251.
- [104] S. Roy, B. Kundu, and D. Chatterjee, "Cloud Based Real-Time Vibration and Temperature Monitoring System for Wind Turbine," *Studies in Infrastructure and Control*. Springer Singapore, pp. 89–99, 2021, doi: 10.1007/978-981-16-1011-0_9.
- [105] M. Tiboni, C. Remino, R. Bussola, and C. Amici, "A Review on Vibration-Based Condition Monitoring of Rotating Machinery," *Applied Sciences*, vol. 12, no. 3, p. 972, 2022, doi: 10.3390/app12030972.
- [106] Y. Arbella-Feliciano, C. A. Trinchet-Varela, L. L. Lorente-Leyva, and D. H. Peluffo-Ordóñez, "Condition Monitoring of Wind Turbines: A Case Study of the Gibara II Wind Farm," *Journal Européen des Systèmes Automatisés*, vol. 56, no. 2, pp. 329–335, 2023, doi: 10.18280/jesa.560218.
- [107] D. Broda, "Description of varying operational conditions in wind turbines," 2011.
- [108] S. Bogoevska, M. Spiridonakos, E. Chatzi, E. Dumova-Jovanoska, and R. Höffer, "A Data-Driven Diagnostic Framework for Wind Turbine Structures: A Holistic Approach," Sensors (Basel, Switzerland), vol. 17, no. 4, p. 720, Mar. 2017, doi: 10.3390/s17040720.
- [109] Y. Vidal, "Artificial Intelligence for Wind Turbine Condition Monitoring," *Energies*, vol. 16, no. 4, p. 1632, 2023, doi: 10.3390/en16041632.
- [110] T. Zhang, R. Dwight, and K. El-Akruti, "Condition Based Maintenance and Operation of Wind Turbines," *Lecture Notes in Mechanical Engineering*. Springer International Publishing, pp. 1013–1025, 2014, doi: 10.1007/978-3-319-09507-3_87.
- [111] W. Qiao, P. Zhang, and M.-Y. Chow, "Condition Monitoring, Diagnosis, Prognosis, and Health Management for Wind Energy Conversion Systems," *IEEE Transactions on Industrial Electronics*, vol. 62, no. 10, pp. 6533–6535, 2015, doi: 10.1109/tie.2015.2464785.
- [112] V. M. Karbhari and L. S.-W. Lee, "Vibration-based damage detection techniques for structural health monitoring of civil infrastructure systems," *Structural Health Monitoring* of Civil Infrastructure Systems. Elsevier, pp. 177–212, 2009, doi: 10.1533/9781845696825.1.177.
- [113] Y. Amirat, V. Choqueuse, and M. Benbouzid, "Condition monitoring of wind turbines based on amplitude demodulation," 2010 IEEE Energy Conversion Congress and Exposition. IEEE, pp. 2417–2421, 2010, doi: 10.1109/ecce.2010.5617914.
- [114] G. T. Zheng and W. J. Wang, "A New Cepstral Analysis Procedure of Recovering Excitations for Transient Components of Vibration Signals and Applications to Rotating Machinery

Condition Monitoring," *Journal of Vibration and Acoustics*, vol. 123, no. 2, pp. 222–229, 2001, doi: 10.1115/1.1356696.

- [115] C. Peeters, P. Guillaume, and J. Helsen, "Vibration-based bearing fault detection for operations and maintenance cost reduction in wind energy," *Renewable Energy*, vol. 116, pp. 74–87, 2018, doi: 10.1016/j.renene.2017.01.056.
- [116] D. Abboud, J. Antoni, S. Sieg-Zieba, and M. Eltabach, "Envelope analysis of rotating machine vibrations in variable speed conditions: A comprehensive treatment," *Mechanical Systems* and Signal Processing, vol. 84, pp. 200–226, 2017, doi: 10.1016/j.ymssp.2016.06.033.
- [117] N. Jamaludin, D. Mba, and R. H. Bannister, "Condition monitoring of slow-speed rolling element bearings using stress waves," *Proceedings of the Institution of Mechanical Engineers, Part E: Journal of Process Mechanical Engineering*, vol. 215, no. 4, pp. 245–271, 2001, doi: 10.1177/095440890121500401.
- [118] P. Waskito, S. Miwa, Y. Mitsukura, and H. Nakajo, "Parallelizing Hilbert-Huang Transform on a GPU," 2010 First International Conference on Networking and Computing. IEEE, pp. 184– 190, 2010, doi: 10.1109/ic-nc.2010.44.
- [119] X. Wan, S. Zhang, J. Zhang, H. Chai, and H. Zhao, "Time-Frequency Analysis of Vibration Signal Distribution of Rotating Machinery Based on Machine Learning and EMD Decomposition," *Lecture Notes of the Institute for Computer Sciences, Social Informatics and Telecommunications Engineering*. Springer International Publishing, pp. 115–128, 2022, doi: 10.1007/978-3-030-94182-6_9.
- [120] E. Filho, A. Santos, H. Sinezio, E. Filho, A. Fernandes, and J. Seixas, "Empirical Mode Decomposition: Theory and Applications in Underwater Acoustics," *Journal of Communication and Information Systems*, vol. 37, no. 1, p. 145, 2022, doi: 10.14209/jcis.2022.16.
- [121] A. R. Nejad *et al.*, "Wind turbine drivetrains: state-of-the-art technologies and future development trends," *Wind Energy Science*, vol. 7, no. 1, pp. 387–411, 2022, doi: 10.5194/wes-7-387-2022.
- [122] A. Fernandez, "Demodulation or envelope analysis," Power-MI, 2020. .
- [123] E. P. Carden and P. Fanning, "Vibration Based Condition Monitoring: A Review," *Structural Health Monitoring*, vol. 3, no. 4, pp. 355–377, 2004, doi: 10.1177/1475921704047500.
- [124] M. Niyazi and A. N. Babadi, "Control and Monitoring of Wind Farms Based on IoT Application for Energy Conversion," *Handbook of Smart Energy Systems*. Springer International Publishing, pp. 1–5, 2023, doi: 10.1007/978-3-030-72322-4_177-1.
- [125] Z. Gao and P. Odgaard, "Real-time monitoring, fault prediction and health management for offshore wind turbine systems," *Renewable Energy*, vol. 218, p. 119258, 2023, doi: 10.1016/j.renene.2023.119258.
- [126] E. Kipchirchir, M. H. Do, J. G. Njiri, and D. Söffker, "Prognostics-based adaptive control strategy for lifetime control of wind turbines," *Wind Energy Science*, vol. 8, no. 4, pp. 575– 588, 2023, doi: 10.5194/wes-8-575-2023.
- [127] M. D. Reder, E. Gonzalez, and J. J. Melero, "Wind Turbine Failures Tackling current Problems in Failure Data Analysis," *Journal of Physics: Conference Series*, vol. 753, p. 72027, 2016, doi: 10.1088/1742-6596/753/7/072027.
- [128] X. Chen, M. A. Eder, A. Shihavuddin, and D. Zheng, "A Human-Cyber-Physical System toward Intelligent Wind Turbine Operation and Maintenance," *Sustainability*, vol. 13, no. 2, p. 561, 2021, doi: 10.3390/su13020561.
- [129] Z. Tian, T. Jin, B. Wu, and F. Ding, "Condition based maintenance optimization for wind power generation systems under continuous monitoring," *Renewable Energy*, vol. 36, no. 5, pp. 1502–1509, 2011, doi: 10.1016/j.renene.2010.10.028.
- [130] D. Goyal, B. S. Pabla, S. S. Dhami, and K. Lachhwani, "Optimization of condition-based maintenance using soft computing," *Neural Computing and Applications*, vol. 28, no. S1, pp. 829–844, 2016, doi: 10.1007/s00521-016-2377-6.
- [131] P. Li, S. L. J. Hu, and H. J. Li, "Noise Issues of Modal Identification using Eigensystem Realization Algorithm," *Procedia Engineering*, vol. 14, pp. 1681–1689, 2011, doi: 10.1016/j.proeng.2011.07.211.
- [132] J. Yang, Y. Sun, H. Jing, and P. Li, "An improved NExT method for modal identification with

tests validation," *Engineering Structures*, vol. 274, p. 115192, 2023, doi: 10.1016/j.engstruct.2022.115192.

- [133] M. Song, N. Partovi Mehr, B. Moaveni, E. Hines, H. Ebrahimian, and A. Bajric, "One year monitoring of an offshore wind turbine: Variability of modal parameters to ambient and operational conditions," *Engineering Structures*, vol. 297, p. 117022, 2023, doi: 10.1016/j.engstruct.2023.117022.
- [134] J.-H. Wan, R. Bai, X.-Y. Li, and S.-W. Liu, "Natural Frequency Analysis of Monopile Supported Offshore Wind Turbines Using Unified Beam-Column Element Model," *Journal of Marine Science and Engineering*, vol. 11, no. 3, p. 628, 2023, doi: 10.3390/jmse11030628.
- [135] J. Valasek and W. Chen, "Observer/Kalman Filter Identification for Online System Identification of Aircraft," *Journal of Guidance, Control, and Dynamics*, vol. 26, no. 2, pp. 347–353, 2003, doi: 10.2514/2.5052.
- [136] W. Yang, P. J. Tavner, C. J. Crabtree, and M. Wilkinson, "Cost-Effective Condition Monitoring for Wind Turbines," *IEEE Transactions on Industrial Electronics*, vol. 57, no. 1, pp. 263–271, 2010, doi: 10.1109/tie.2009.2032202.
- [137] M.-Q. Tran, Y.-C. Li, C.-Y. Lan, and M.-K. Liu, "Wind Farm Fault Detection by Monitoring Wind Speed in the Wake Region," *Energies*, vol. 13, no. 24, p. 6559, 2020, doi: 10.3390/en13246559.
- [138] S. Y. Oh et al., "Condition-based maintenance of wind turbine structures: A state-of-the-art review," Renewable and Sustainable Energy Reviews, vol. 204, p. 114799, 2024, doi: 10.1016/j.rser.2024.114799.
- [139] R. A. Swartz, J. P. Lynch, S. Zerbst, B. Sweetman, and R. Rolfes, "Structural monitoring of wind turbines using wireless sensor networks," *Smart Structures and Systems*, vol. 6, no. 3, pp. 183–196, 2010, doi: 10.12989/sss.2010.6.3.183.
- [140] R. Niyirora, W. Ji, E. Masengesho, J. Munyaneza, F. Niyonyungu, and R. Nyirandayisabye, "Intelligent damage diagnosis in bridges using vibration-based monitoring approaches and machine learning: A systematic review," *Results in Engineering*, vol. 16, p. 100761, 2022, doi: 10.1016/j.rineng.2022.100761.
- [141] S. Sahoo, K. Kushwah, and A. K. Sunaniya, "Health Monitoring of Wind Turbine Blades through Vibration Signal Using Advanced Signal Processing Techniques," 2020 Advanced Communication Technologies and Signal Processing (ACTS). IEEE, pp. 1–6, 2020, doi: 10.1109/acts49415.2020.9350405.
- [142] W. Hernandez, J. L. Maldonado-Correa, and A. Méndez, "Frequency-domain analysis of performance of a wind turbine," *Electronics Letters*, vol. 52, no. 3, pp. 221–223, 2016, doi: 10.1049/el.2015.2711.
- [143] M. Islam, H. A. Mohammadpour, P. Stone, and Y.-J. Shin, "Time-frequency based power quality analysis of variable speed wind turbine generators," *IECON 2013 - 39th Annual Conference of the IEEE Industrial Electronics Society*. IEEE, pp. 6426–6431, 2013, doi: 10.1109/iecon.2013.6700194.
- [144] W. Xin, Y. Liu, Y. He, and B. Su, "Amplitude envelope analysis for feature extraction of directdriven wind turbine bearing failure," *Proceedings of the 10th World Congress on Intelligent Control and Automation*. IEEE, pp. 3173–3176, 2012, doi: 10.1109/wcica.2012.6358418.
- [145] H. Toubakh, M. Sayed-Mouchaweh, and E. Duviella, "Advanced Pattern Recognition Approach for Fault Diagnosis of Wind Turbines," 2013 12th International Conference on Machine Learning and Applications. IEEE, pp. 368–373, 2013, doi: 10.1109/icmla.2013.150.
- [146] K. F. Abdulraheem and G. Al-Kindi, "Wind Turbine Condition Monitoring using Multi-Sensor Data system," *International Journal of Renewable Energy Research*, no. v8i1, 2018, doi: 10.20508/ijrer.v8i1.6506.g7276.
- [147] M. J. Ali, A. Mondal, and P. Dutta, "Intelligent Monitoring and Control of Wind Turbine Prototype Using Internet of Things (IoT)," 2022 IEEE International IOT, Electronics and Mechatronics Conference (IEMTRONICS). IEEE, pp. 1–6, 2022, doi: 10.1109/iemtronics55184.2022.9795845.
- [148] J. Cuesta, U. Leturiondo, Y. Vidal, and F. Pozo, "Challenges on prognostics and health management for wind turbine components," *Journal of Physics: Conference Series*, vol. 2745, no. 1, p. 12003, 2024, doi: 10.1088/1742-6596/2745/1/012003.

- [149] N. O. Farrar, M. H. Ali, and D. Dasgupta, "Artificial Intelligence and Machine Learning in Grid Connected Wind Turbine Control Systems: A Comprehensive Review," *Energies*, vol. 16, no. 3, p. 1530, 2023, doi: 10.3390/en16031530.
- [150] J. Kang, Z. Wang, and C. Guedes Soares, "Condition-Based Maintenance for Offshore Wind Turbines Based on Support Vector Machine," *Energies*, vol. 13, no. 14, p. 3518, 2020, doi: 10.3390/en13143518.
- [151] S. Oh and T. Ishihara, "On the parameter sensitivity in structural parameter identification using Eigensystem Realization Algorithm for a MW-size wind turbine," *Journal of Physics: Conference Series*, vol. 1037, p. 52026, 2018, doi: 10.1088/1742-6596/1037/5/052026.
- [152] Y. Liu, J.-M. Zhang, Y.-T. Min, Y. Yu, C. Lin, and Z.-Z. Hu, "A digital twin-based framework for simulation and monitoring analysis of floating wind turbine structures," *Ocean Engineering*, vol. 283, p. 115009, 2023, doi: 10.1016/j.oceaneng.2023.115009.
- [153] M. Alswaitti, M. K. Ishak, and N. A. M. Isa, "Optimized gravitational-based data clustering algorithm," *Engineering Applications of Artificial Intelligence*, vol. 73, pp. 126–148, 2018, doi: 10.1016/j.engappai.2018.05.004.
- [154] J. Naik, A. Acharya, and J. Thaker, "Revolutionizing condition monitoring techniques with integration of artificial intelligence and machine learning," *Materials Today: Proceedings*, 2023, doi: 10.1016/j.matpr.2023.08.262.
- [155] B. A. Tama, M. Vania, S. Lee, and S. Lim, "Recent advances in the application of deep learning for fault diagnosis of rotating machinery using vibration signals," *Artificial Intelligence Review*, vol. 56, no. 5, pp. 4667–4709, 2022, doi: 10.1007/s10462-022-10293-3.
- [156] N. P. Chavan, M. S. Kale, S. P. Deshmukh, and C. M. Thakar, "A Review on Development and Trend of Condition Monitoring and Fault Diagnosis," *ECS Transactions*, vol. 107, no. 1, pp. 17863–17870, 2022, doi: 10.1149/10701.17863ecst.
- [157] W. Yang, Z. Wang, and Y. Choy, "Prediction of sound radiation from an unbaffled long enclosure with the ground," *Mechanical Systems and Signal Processing*, vol. 149, p. 107232, 2021, doi: 10.1016/j.ymssp.2020.107232.
- [158] F. Xiao, C. Tian, I. Wait, Z. (Joey) Yang, B. Still, and G. S. Chen, "Condition monitoring and vibration analysis of wind turbine," *Advances in Mechanical Engineering*, vol. 12, no. 3, p. 168781402091378, 2020, doi: 10.1177/1687814020913782.
- [159] A. Al-Habaibeh, A. Boateng, and H. Lee, "Innovative Strategy for Addressing the Challenges of Monitoring Off-Shore Wind Turbines for Condition-Based Maintenance," Springer Proceedings in Energy. Springer International Publishing, pp. 189–196, 2021, doi: 10.1007/978-3-030-63916-7_24.
- [160] Z. Wu, Y. Li, and P. Wang, "A hierarchical modeling strategy for condition monitoring and fault diagnosis of wind turbine using SCADA data," *Measurement*, vol. 227, p. 114325, 2024, doi: 10.1016/j.measurement.2024.114325.
- [161] M. Schütt, "Wind Turbines and Property Values: A Meta-Regression Analysis," Environmental and Resource Economics, vol. 87, no. 1, pp. 1–43, 2023, doi: 10.1007/s10640-023-00809-y.
- [162] C. F. de Lima Munguba *et al.*, "Ensemble learning framework for fleet-based anomaly detection using wind turbine drivetrain components vibration data.," *Engineering Applications of Artificial Intelligence*, vol. 133, p. 108363, 2024, doi: 10.1016/j.engappai.2024.108363.
- [163] D. Ferguson and V. M. Catterson, "Big Data Techniques for Wind Turbine Condition Monitoring," 2014.
- [164] N. V Poorima, B. Srinivasan, and S. Karthikeyan, "Monitoring the Wind Turbine Condition Using Big Data Technique," Asian Journal of Computer Science and Technology, vol. 8, no. S1, pp. 98–102, 2019, doi: 10.51983/ajcst-2019.8.s1.1941.
- [165] C. Q. Gómez, M. A. Villegas, F. P. García, and D. J. Pedregal, "Big Data and Web Intelligence for Condition Monitoring," Advances in Data Mining and Database Management. IGI Global, pp. 149–163, 2015, doi: 10.4018/978-1-4666-8505-5.ch008.
- [166] N. Amuthan, M. B. M, P. Velrajkumar, N. Sivakumar, and T. Jarin, "IOT based adjustment mechanism for direct reference model adaptive IMC to support voltage sag in DFIG wind farm," *Measurement: Sensors*, vol. 27, p. 100809, 2023, doi:

10.1016/j.measen.2023.100809.

- [167] D. Chan and J. Mo, "Life Cycle Reliability and Maintenance Analyses of Wind Turbines," Energy Procedia, vol. 110, pp. 328–333, 2017, doi: 10.1016/j.egypro.2017.03.148.
- [168] E. Artigao, S. Martín-Martínez, A. Honrubia-Escribano, and E. Gómez-Lázaro, "Wind turbine reliability: A comprehensive review towards effective condition monitoring development," *Applied Energy*, vol. 228, pp. 1569–1583, 2018, doi: 10.1016/j.apenergy.2018.07.037.
- [169] D. Astolfi, F. De Caro, and A. Vaccaro, "Condition Monitoring of Wind Turbine Systems by Explainable Artificial Intelligence Techniques," *Sensors (Basel, Switzerland)*, vol. 23, no. 12, p. 5376, Jun. 2023, doi: 10.3390/s23125376.
- [170] E. Bechhoefer, M. Wadham-Gagnon, and B. Boucher, "Initial Condition Monitoring Experience on a Wind Turbine," *Annual Conference of the PHM Society*, vol. 4, no. 1, 2012, doi: 10.36001/phmconf.2012.v4i1.2119.
- [171] G. Marsh and D. Robb, "Patently innovative," *Refocus*, vol. 8, no. 2, pp. 30–35, 2007, doi: 10.1016/s1471-0846(07)70047-1.
- [172] F. K. Moghadam and A. R. Nejad, "Online condition monitoring of floating wind turbines drivetrain by means of digital twin," *Mechanical Systems and Signal Processing*, vol. 162, p. 108087, 2022, doi: 10.1016/j.ymssp.2021.108087.
- [173] M. Fahim, V. Sharma, T.-V. Cao, B. Canberk, and T. Q. Duong, "Machine Learning-Based Digital Twin for Predictive Modeling in Wind Turbines," *IEEE Access*, vol. 10, pp. 14184– 14194, 2022, doi: 10.1109/access.2022.3147602.
- [174] P. Tchakoua, R. Wamkeue, F. Slaoui-Hasnaoui, T. A. Tameghe, and G. Ekemb, "New trends and future challenges for wind turbines condition monitoring," 2013 International Conference on Control, Automation and Information Sciences (ICCAIS). IEEE, pp. 238–245, 2013, doi: 10.1109/iccais.2013.6720561.
- [175] P. S. Thakur, "Deficiency Diagnosis and Conditional Monitoring Way of a Wind Turbine," International Journal of Advanced Research in Electrical, Electronics and Instrumentation Engineering, vol. 5, no. 5, pp. 4296–4301, 2016, doi: 10.15662/IJAREEIE.2016.0505153.